Title: Holistic overview of natural ventilation and mixed mode in built environment of warm climate zones and hot seasons

Authors:

David Bienvenido-Huertas¹, María Luisa de la Hoz-Torres^{1*}, Antonio J. Aguilar¹, Blanca Tejedor², Daniel Sánchez-García³

¹Department of Building Construction, University of Granada. Granada, Spain

²Department of Project and Construction Engineering, Universitat Politècnica de Catalunya (UPC). Terrassa (Barcelona), Spain

³Department of Electrical Engineering, University Carlos III of Madrid, Madrid, Spain

* <u>mlhoz@ugr.es</u> - Author to whom correspondence should be addressed.

Abstract:

The climate change leads to periods of extreme events (i.e. reduction of cold seasons, heat waves, overheating, urban heat island among others) that affect the performance of residential and tertiary buildings with high occupancy (i.e. hospitals, schools, commercial centres, offices etc). However, most of low-carbon policies do not consider the ventilation as a mitigation measure. In fact, a lack of studies on natural ventilation (NV) and mixed-mode (MM) strategies was detected, especially for warm regions or areas with hot and humid climates. This paper aims to carry out a bibliometric analysis from 1928 to 2023, to observe the evolution of the topic. After identifying the main research clusters (thermal comfort, energy efficiency, indoor air quality and simulation tools) by science mapping, the most relevant publications of the last 20 years were assessed (2003 – 2023). The results of this study revealed that only 1.51% of the scientific documents in 95 years corresponded to an extensive literature review, although epidemic or disease outbreaks led to peaks of production in this topic. This emphasizes the importance of observing what was done and how was implemented over the years. Regarding the clusters, some relevant aspects can be highlighted: (i) non-homogeneity of studies on NV or MM related to building type; (ii) interregional projects should be drawn up to check the effectiveness of NV and MM, especially when other architectural techniques are adopted (i.e. solar chimneys, window wall ratio –WWR-, thermally activated building structures –TABS- etc); (iii) the optimization of simulation tools should be based on the incorporation of BIM and generative design for NV and MM.

Keywords:

natural ventilation; mixed mode; thermal comfort; energy efficiency; built environment; IAQ; software; simulation

Highlights:

- Only 1.51% of scientific production on ventilation corresponds to review journal papers.
- Four clusters were identified: thermal comfort, IAQ, energy efficiency and simulation.
- Heterogeneity of studies related to building type.
- NV and MM strategies are effective in achieving reductions in energy consumption.
- An holistic analysis of NV and MM with several energy saving measures is required.
- Seasonal variations on IAQ have been scarcely assessed in the scientific literature.
- BIM and generative design for NV and MM should be adopted to optimize simulation tools.

1. Introduction

The planet Earth is suffering a process of warming due to climate change [1,2]. This causes effects as varied as the extension of summers, desertification and the reduction of cold seasons [3]. This increase in temperatures constitutes a challenge for survival. In the case of buildings, the majority of low-carbon policies propose two mitigation measure domains: (i) the reduction of environmental impact (minimizing energy consumption), and (ii) the adaptation of users to guarantee adequate conditions of thermal comfort [4]. For this reason, the policies are mainly based on the technological improvement of buildings [5]. However, energy saving and thermal adaptability techniques used in the past are often forgotten.

In this sense, a measure that can have a positive impact on both mentioned aspects is ventilation. Ventilation is a measure traditionally applied to enhance indoor air quality (IAQ). Nevertheless, the importance of ventilation to reduce energy-cooling demand has only been highlighted in the recent years [6]. Natural ventilation (NV) is characterized by depending on the external conditions, since it usually occurs due to pressure gradient and stack effect. Given the variability of cases, the following classification of NV can be established [7]: (i) unilateral ventilation; (ii) cross ventilation; and (iii) stack ventilation. Figure 1 schematically shows the principles of these classes of NV.



NV: Stack ventilation

Figure 1. Natural ventilation schemes.

Despite of being a cost-effective measure [8,9], NV has several limitations. For example, it can be difficult to ensure users' thermal comfort [10]. Given this circumstance, mixed-mode (MM) could be an alternative. MM controls the temperature of the indoor air with NV and air conditioning. The use of one system or another depends on the outside temperature conditions [11]: when the outside temperature has adequate conditions for thermal comfort, the windows are opened; otherwise, the air conditioning will be used. MM makes it possible to ensure a balance between maintaining thermal comfort and adequate use of air conditioning.

1.1. Aim of this study

Thus, both NV and MM can be effective measures to improve the situation of buildings in the face of climate change. As a result, a wide variety of studies have been developed in the scientific literature. However, no previous review studies were detected to comprehensively address the application of NV and MM in warm regions (i.e. analysis of the influence on energy, thermal comfort, indoor air quality, control strategies, etc.). Most of the existing reviews have carried out partial analyses of these aspects and, in some cases, without specifying the climatic zone: (i) Kim and de Dear [12] analysed the state of the art in relation to MM and its influence on thermal comfort and energy saving; (ii) Mateus et al. [13] reviewed previous researches focused on the characterization of the ventilation of large air masses; (iii) Zhong et al. [14] evaluated single-sided natural ventilation, highlighting aspects such as methodologies, classifications, and influence factors; (iv) Ferrari et al. [15] reported past studies of classroom ventilation to reduce virus transmission; (v) Izadyar and Miller [16] carried out an extensive literature review about the influence of ventilation typologies on the risk of contagion; (vi) Ahmed et al. [17] analysed prior studies that combine natural ventilation and solar chimneys in hot areas, considering the domain of thermal comfort; and (vii) recently, Jiang et al. [18] reviewed the methods for modelling cross ventilation. As observed, it is necessary to know the current state of the art about the influence of cost-effective ventilation strategies as a measure against climate change. Based on this, the aim of this work was to provide a detailed framework on the applicability of NV or MM in areas with warm climates, as well as regions under singular effects such as urban heat islands (UHI) or heat waves (HW).

For this holistic overview, the manuscript was structured in five sections. Section 2 describes the methodology used in the research. Section 3 reports the bibliometric analysis related to the topic. Section 4 presents the assessment of the research clusters detected in the scientific literature (thermal comfort, energy efficiency, indoor air quality and simulation tools). Finally, Section 5 reports the discussions and conclusions of the study.

2. Methodology

The research methodology was divided into two steps, as shown in Figure 2.



Figure 2. Flowchart of the research methodology (Source: Elaborated by the authors).

Firstly, the bibliometric analysis was conducted, considering a time span of 95 years (from 1928 to 2023) and adopting Scopus database as reference. Several authors stated that Scopus has an expanded spectrum of scientific publications, covering up 20% more than Web of Science [19–21]. In addition, the data statistics from Scopus offers the possibility to know the geographic distribution of the publications, the type of document and the analysis of the authors. Subsequently, science mapping was applied to identify hot topics and trends, since it is a quantitative approach which allows to classify

and to evaluate bibliographic networks in a specific research field [22,23]. For this purpose, the most widely used opensource software (VOSviewer) was executed [20,24,25]. Secondly, taking into account the results of the first step of the methodology, an evaluation of research clusters was carried out for the last 20 years (from 2003 to 2023). This implied the review of the most relevant studies on: (i) thermal comfort and ventilation in warm climate zones; (ii) implications of the ventilation in the energy performance of buildings; (iii) indoor air quality and ventilation in warm climate zones; (iv) simulation tools for ventilation.

3. Bibliometric analysis

3.1. Data statistics from Scopus database

According to Section 2, it was needed to develop a literature retrieval and a search refinement. As shown in Table 1, the combination of keywords was computed among three categories: (i) ventilation strategy; (ii) climate and extreme thermal events; (iii) type of buildings. Considering that the first article on this research field corresponds to 1928, 1856 scientific publications were found for a period of 95 years. Indeed, it can be reported that 61.6% of documents corresponded to journal papers (Figure 3), while 38.4% were attributed to grey literature (i.e. book chapters, conference papers and so on). However, only 1.5% referred to review articles (28 documents in 95 years). Concerning the geographic distribution (Figure 4), the ten countries with a higher number of publications were China (260), UK (260), USA (180), Japan (120), India (109), Italy (91), Spain (80), Australia (78), Malaysia (74) and Germany (69). All of them represents the 71.2% of the analysed sample. It should be noted that the common peaks of research production were given when an epidemic or disease outbreak had occurred: (i) the avian and swine influenza (2000 – 2009) [26]; (ii) the West African Ebola epidemic (2013 – 2016) [27]; and (iii) the Covid-19 pandemic (2019 – present) [28].

Table 1. Queries used for categories 1 to 3.

	Category	Query		
1	Ventilation strategy	TITLE-ABS-KEY("natural ventilation" OR "naturally ventilated" OR "NV" OR "mixed-mode" OR "MM" OR "free running" OR "FR" OR "unilateral ventilation" OR "cross ventilation" OR "stack ventilation")		
2	Climate & extreme events	TITLE-ABS-KEY("summer" OR "warm" OR "hot" OR "Mediterranean" OR "humid climate" OR "climate change" OR "urban heat island" OR "UHI" OR "heat wave" OR "HW" OR "cooling systems" OR "passive cooling" OR "free cooling")		
3	Building type	TITLE-ABS-KEY("dwelling" OR "house" OR "residential building" OR "historic building" OR "heritage building" OR "office" OR "shopping center" OR "shopping mall" OR "commercial building" OR "educational building" OR "schools" OR "universities" OR "hospital" OR "clinic" OR "hospice" OR "nursing homes" OR "elderly care centers" OR "ECC")		



Figure 3. Document type distribution -from 1928 to 2023- (Source: Elaborated by the authors).



Figure 4. Publications for the top 10 countries -from 1928 to 2023- (Source: Elaborated by the authors).

Regarding the ten researchers with more influence on this topic, these are listed in Table 2. However, most of them did not collaborate together.

Researcher	Number of documents	University (Country)	h-index
Dr. Tetsu Kubota	21	Hiroshima University (Japan)	14
Dr. Hom B. Rijal	19	Tokyo City University (Japan)	30
Dr. Roberto Lamberts	12	Federal University of Santa Catarina (Brazil)	34
Dr. Kevin J. Lomas	12	Loughborough University (UK)	37
Dr. Richard De Dear	12	University of Sydney (Australia)	63
Dr. Madhavi Indraganti	10	Qatar University (Qatar)	19
Dr. Maria Kolokotroni	10	Brunel University London Uxbridge (UK)	30
Dr. Anna Mavrogianni	10	University College London (UK)	31
Dr. Andreas Wagner	10	Karlsruhe Institute of Technology (Germany)	29
Dr. Runming Yao	10	Chongqing University (China)	38

Table 2. The ten researchers with more influence on this topic.

3.2. Science mapping: overview of the research field

After conducting science mapping with VOSviewer, the co-occurrence network of keywords for a time span of 95 years (1928 – 2023) and the overlay visualization were plotted (Figures 5 and 6). As a result, a total of 288 items and 18952 links were obtained. From 1856 scientific publications, and applying a minimum occurrence of 15 times, 5 macro research areas were detected (Figure 5). Cluster 1 (energy efficiency; red colour) covers 99 items related to: building codes, architectural design (passive and active strategies), building energy performance in front of climatic conditions, energy management, intelligent buildings, retrofitting (i.e. improvement of thermal insulation of envelopes), building life cycle, and investments. Cluster 2 (indoor air quality - green colour) with 63 items emphasizes the air pollution and how this can be monitored and controlled (i.e. airflow, ventilation rates, type of pollutants –CO₂, PM-, urbans areas, risk assessment), especially in educational buildings. Cluster 3 (indoor thermal comfort - blue colour) covers 59 items that mainly refer to users' behaviour and adaptive comfort models, considering the data gathered from field surveys. Cluster 4 (simulation tools for ventilation - yellow colour) with 55 items involves existing computational simulation tools to develop numerical models. This allows evaluating wind effects, ventilation performance (during the day and the night), stack effect, thermal buoyancy, heat fluxes, cooling loads, solar chimneys and so on. Cluster 5 (hybrid ventilation systems - purple colour), with only 12 items, is more focused on the use of hybrid ventilation systems. In particular, those facilities designed for hospitals or neonatal spaces.

Furthermore, the findings revealed a strong relationship among Cluster 1, 2 and 4. By way of example, the implementation of ventilation models (cluster 4) could help to achieve a better use of energy (cluster 1) and to reduce carbon dioxide levels (cluster 2) inside buildings.



Figure 5. Co-occurrence network of keywords (Source: Created by the authors using VOSviewer and Scopus data).

Concerning the overlay visualization (Figure 6), it can be extrapolated that the first studies (before 2013) were associated to the application of computational fluid dynamics (CFD) simulations to assess air pollution of offices or to evaluate the building envelope of houses. The different scenarios were defined by season (summer / winter), building characteristics (i.e. heat transfer of construction materials) and gender population (male / female). From 2013 to 2016, the researchers concentrated their efforts on analysing the indoor air quality and thermal comfort, without connection between these two domains. Nevertheless, the first on-site experimental campaigns with microclimate stations and surveys about thermal sensation were carried out in this period, taking hospitals or schools as case studies. Regarding data processing, regression analysis was chosen as calculation method. In addition, several passive strategies like solar chimneys, vernacular architecture (use of local materials) or heat storage systems were tested as a solution related to natural ventilation. From 2016 to the present, with the effects of climate change and covid-19, the interest on zero energy buildings or smart buildings and ventilation systems increased. According to the Scopus database, it is estimated that 53.64% of the total scientific documents were published in the last 7 years. The current trends are: the influence of indoor air pollutants in humans; the use of phase change materials for greater sustainability; historic preservation of patrimonial assets when opting for NV; the impact of overheating periods and seasonal variations on the performance of ventilation systems; investments and energy savings associated to ventilation systems; and the implementation of adaptive models to achieve both operative and comfort temperatures. Actually, simulations tools to unify thermal comfort and ventilation domains are still on going.



Figure 6. Overlay visualization (Source: Created by the authors using VOSviewer and Scopus data).

4. Assessment of research clusters

4.1. Thermal comfort and ventilation in warm climate zones

Thermal comfort is defined as the condition that expresses satisfaction with the surroundings and is assessed by subjective evaluation [29]. Thermal sensation is mainly related to the thermal balance of the human-body as a whole. This balance is influenced by physical factors (i.e. activity and clothing) and environmental parameters (i.e. air temperature, mean radiant temperature, air velocity and air humidity) [30]. Most of the studies on thermal comfort are mainly based on static models (i.e. PMV-PPD [30], SET [29], etc.) or adaptive models (ASHRAE-55, EN 16798-1:2019) (Figure 7). Static thermal comfort models usually consider steady-state heat transfer conditions between the human body and the environment, assigning a comfort vote (e.g. PMV). In contrast, adaptive models take into account the human responses and relative changes to the indoor environment. In this sense, physiological acclimatisation, psychological expectation and behavioural factors can influence the human thermal perception [31].



Figure 7. Timeline of thermal comfort models (Source: elaborated by the authors).

ASHRAE-55 and EN 16798-1 standards establish the use of adaptive models in occupant-controlled naturally conditioned spaces. These models tend to have a wider comfort temperature band, which could lead to significant energy savings in naturally ventilated (NV), air-conditioned (AC) and mixed mode (MM) buildings [32].

Current thermal comfort standards still differentiate between NV and AC buildings, while buildings that alternate between AC and NV (i.e. MM) are often left in an undefined range [33,34]. Several studies analysed how NV strategies influence thermal comfort in buildings. A summary of these works is shown in Table 3, regardless building type (most studies do not indicate whether the NV strategy is single-side or cross-ventilation). Along this line, Figure 8 shows a summary of the neutral temperature distribution of the studies reviewed in the scientific literature. The values show that the occupants of residential and offices NV buildings have a higher tolerance to thermal environment at higher temperatures, while the values show lower neutral temperature for the NV+F, MM and FR buildings. In contrast, a higher tolerance to thermal environment is observed in educational buildings with NV+F and MM systems. It should be noted that the occupants' density, mode of operation, dress code and space characteristics of rooms in educational buildings may vary from that of residential or office buildings, and therefore, influence occupants' thermal preference. Regarding nursing buildings, Figure 8 shows the lowest neutral temperature. This difference may be due to the fact that the residents of such buildings are older people and have different characteristics (lower metabolic rates, wear more insulated clothing, etc.) than a young adults. In addition, in this type of building, users often have disabilities and pathologies that can significantly affect thermal comfort. Finally, a wider range of neutral temperatures is observed in the different building types for the NV systems compared to the others.

Year	Country	Climate	Ventilation system	Type of building	Equation	Acceptability/ Comfort limits	Reference
2023	India	Aw	NV + CF	Educational	$TSV = 0.37 T_{op} - 10.94$	27.0 – 32.4 °C	[35]
2023	India	Cwa	NV + CF	Residential	$TSV = 0.09T_g - 2.10$ T _{comf} = 0.54T _{out} + 12.32	12.2 – 34.4 °C [-1 +1]	[36]
2022	Zanzibar	Aw	NV	Residential	$TSV = 0.65 T_{op} - 17.74$	26.0 – 28.9 °C [-1 +1]	[37]
2022	India	Aw	NV + CF	Educational	MTSV=0.41*Top - 11.81	25.0 – 32.5°C	[38]
2022	India	BSh – Aw – Cwa – Cwb	ММ	Residential		18.2 – 38.3 °C	[39]
					$MTSV = 0.0997T_{op} - 1.8058$	8.1 – 28.1 °C	
2022	Spain	Csa – Csb	NV	Nursing home			[40]
	Portugal			-	$MTSV = 0.1235T_{op} - 1.7371$	6.0 – 22.2 °C	
2022	Chana	BSh-Aw	NV - E	Office		23.4 – 28.4 °C	[41]
2022	Ghana	D3II-AW	NV + F			23.1 – 28.1 °C	[41]
2021	Cyprus	BSh	NV	Educational		24.3 - 30.3 °C	[42]
					$MTSV = 0.12T_{op} - 2.8$		
2021	Spain	Csa	NV	Nursing home			[43]
					$MTSV = 0.32 T_{op} - 6.95$		
					$TSV = 0.216 T_{op} - 6.8133$		F 4 43
2021	South Algeria	BWh	NV	Residential	— 0.00 — 0.04 —		[44]
2024	01 :	00		P1 1	$T_{\rm comf} = 0.33 T_{\rm out} + 20.15$		[4]
2021	China	Cfa	NV	Educational	$TSV = 0.289T_{op} - 7.569$	~29.6 (upper limit)	[45]
2021	Japan	Cfa	NV	Educational	$TSV = 0.38 T_{op} - 9.93$	23.3-28.5	[46]
2021	Ecuador	Cfb – Aw – Af	FR MM	Educational	$\begin{split} TSV &= 0.2933 \ T_{op} - 6.4074 \\ TSV &= 0.4985 \ T_{op} - 12.972 \end{split}$	±1.7 °C (90%) ±3.4 °C (80%) ±1.2 °C (90%) ±2.4 °C (80%)	[47]
2021	Nigeria	Am	NV	Educational	$TSV = 0.24T_{op} - 6.90$	25.2 –32.3 °C	[48]
2020	Bangladesh	Aw	NV	Educational	$TSV = 0.14T_{op} - 2.7084$	21.6 – 34 °C	[49]
2020	Daligiauesii	Aw	INV	Euucationai	$T_{\rm comf} = 0.38T_{\rm rm} + 16.10$	21.0 - 54 C	[49]
2020	UK	Cfb	NV	Educational		23.0 – 25.0 °C	[50]
2019	Australia	Csa	NV	Residential	$T_{comf} = 0.27 T_{rm} + 19.2$		[51]
2019	California	Csb	NV + CF	Office		18 – 27 °C	[52]
2019	India	BSh - Cwa	NV + F	Educational	TSV=0.17T _{op} -4.95	16 – 33.7 °C	[53]
2019	China	Cwa	MM	Office	$TSV = 0.18T_{op} - 4.86$	24 - 30 °C (80%)	[54]
					$AC(B1) - TSV = 0.15T_{op} - 4.27$		
					$NV(B1) - TSV = 0.16T_{op} - 4.49$		
					$T_{comf} = 0.16 T_{pma(out)} + 22.58$		
					$AC(B2) - TSV = 0.28T_{op} - 7.15$		
2019	Australia	Cfa	MM	Office	$NV(B2) - TSV = 0.25T_{op} - 6.53$		[55]
2019	Australia	olu	141141	onice	T_{comf} = 0.48 $T_{pma(out)}$ + 15.86		[33]
					AC(B3) - TSV= 0.24Top -7.22		
					NV(B3) - TSV= 0.18T _{op} -4.79		
					$T_{comf} = 0.23 T_{pma(out)} + 20.75$		
2019	China	Cfa	NV + Fan	Residential	$TSV = 0.33T_{op} - 8.61$		[56]
					$TSV = 0.324T_{op} - 8.30$		_
2019	Mexico	Aw	NV	Educational		22.5 – 28.7 °C	[57]
					$T_{comf} = 0.32T_{rm} + 18.45$		
2019	Singapore	Af	NV + AC NV	Educational	$TSV = 0.20 T_{op} - 5.92$	25.2 – 33.7 °C (80%) 27.0 – 31.9 °C (90%)	[58]
2019					$TSV = 0.18 T_{op} - 4.92$		
					10v = 0.10 1 op = 1.72	23.0 – 32.6 °C (80%)	

						24.9 - 30.6 °C (90%)	
2018	India	BSh	NV	Educational	TSV=0.19 T _a -5.04	21.8–31.8 °C	[59]
2018	China	Cfa	NV	Residential	$TSV = 0.2347T_{op} - 6.5646$	22.0 °C – 30.1 °C (80%)	[60]
2018	Indonesia	Am	NV	Educational	$TSV = 0.175T_{op} - 5.074$		[61]
2018	India	BSh	NV + CF	Educational	$TSV = 0.19T_a - 5.04$		[62]
2010	mula	D3II	NV I CI	Lucational	$T_{\rm comf} = 0.49T_{\rm rm} + 13.8$		[02]
2017	China	Cfa – Cwa	FR	Residential	$TSV = 0.155 T_a - 3.76$ $T_{n.d} = 0.29 T_{rm} - 19.90$		[63]
2017	Giillia	Cia – Civa	IK	Residential	$T_{n,d} = 0.29 T_{rm} + 8.25$ $T_{n,d} = 0.709 T_{rm} + 8.25$		[03]
						(Indonesia) FR = 26.7 °C	
2016	Indonesia	Am - Cfa	FR, MM	Office		MM = 27.4 °C	[64]
	Japan					(Japan)	
2016	Japan	Cfa	FR	Educational	$TSV = 0.491T_{op} - 13.1$	FR = 26.6 °C	[65]
2010	Japan			Luucationai	$\frac{13V = 0.4911_{op} = 13.1}{TSV = 0.19T_{op} = 4.82}$		[03]
2016	India	BSh – Aw – Cwa – Cwb	NV MM	Office	$TSV = 0.15T_{op} - 4.02$		[66]
	India	BSh	NV + F	Institutes, offices,	$15V = 0.151_{\text{op}} - 4.02$		[67]
2015					$TSV = 0.299T_a - 8.788$		
				and hostels			
	Japan	Cfa	NV	Residential	$TSV = 0.192T_a - 0.872$		[68]
2014							
					$TSV = 0.189T_g - 0.809$		
2014	India	Aw	NV	Office	$TSV = 0.26T_g - 7.09$		[69]
					$T_{\rm comf} = 0.26 T_{\rm rm} + 21.4$		
0010	·			0.00	$TSV = 0.326 T_g - 8.489$		[= 0]
2013	India	Aw	NV	Office	TOV 0 172 T 4 000		[70]
					$TSV = 0.173 T_g - 4.099$		
0010		DO		5 11 11	$TSV = 0.1155T_a - 3.1715$		[= 4]
2013	Venezuela	BSh	NV + PC	Residential			[71]
2012	Australia	Cfa	ММ	Office	$\frac{\text{TSV} = 0.6445\text{T}_{a} - 18.523}{\text{TSV} = 0.0725\text{T}_{a} - 1.21}$		[24]
2012 2011	India		NV	Residential	$TSV = 0.0735T_{op} - 1.31$	31.38 – 32.9 °C	[34]
		Aw			TSV = 0.21T 0.04		
2010 2008	India China	BSh Cfa	NV NV	Residential Office	$\frac{\text{TSV} = 0.31\text{T}_{\text{g}} - 9.06}{\text{TSV} = 0.32\text{T}_{\text{op}} - 9.12}$	<u>26.0 °C - 32.5 °C (-1 to +1)</u> 25.0 - 31.6 °C	[73-76] [77]
		Af			13v - 0.321 op - 9.12		
2003	China	AI	NV	Educational		27.1 – 29.3 °C	[78]

CF: Ceiling fans; F: fans; PC: passive cooling; MTSV: mean thermal sensation vote; T_a : air temperature; T_{comf} : comfort temperature; T_g : globe temperature; $T_{n,d}$: daily neutral temperature; T_{op} : operative temperature; T_{out} : outdoor temperature; $T_{pma(out)}$: prevailing mean outdoor temperature; T_{rm} : running mean temperature; TSV: thermal sensation vote; B1: Building 1; B2: Building 2; B3: Building 3; FR: free-running;



Figure 8. Summary of neutral temperature in NV and MM buildings in scientific literature (Source: Elaborated by the authors).

4.1.1. Strategies in NV buildings

Numerous studies evaluated the effect of NV strategies in different types of buildings (e.g. hospitals, schools, offices, dwellings or commercial buildings) to check if indoor thermal comfort could be acceptable in warm climates or during summer season. Regarding NV residential buildings, Zepeda-Gil and Natarajan [79] evaluated thermal comfort in NV dwellings in the central Mexican plateau and found that 83% of the comfortable votes were obtained during spring and summer compared to 47.5% in winter.

The study conducted by Ramos et al. [80] in Brazilian residential buildings showed a massive preference for NV environment in all climates zones. In fact, the most common method of adaptation on hot days was opening windows and

doors. Morey et al. [81] investigated overheating risk in FR social housing properties located in central England (UK). Among their major findings, it was detected that a higher proportion of bedrooms exceeded static overheating thresholds, while very few houses exceeded the criteria for thermal comfort adaptive methods. The results reported by Morey suggest that English Midlands housing stock might be expected to overheat significantly in future years, according to static criteria. The results reported in studies conducted in NV nursing homes in Mediterranean climate (SPAIN) supports the use of NV at broader setpoint temperatures without compromising occupants' comfort and well-being [40,43].

Regarding educational buildings, ventilation strategies and indoor thermal conditions should be especially considered given the potential impact on students' performance and health. The study carried out by Wang et al. [45] in China revealed that special attention shall be paid in NV classrooms to avoid summer overheating. In this line, Buonocore et al. [82] also highlighted that thermal discomfort due to heat is an important issue in NV classrooms under hot and humid conditions in Brazil (indoor temperatures above 29 °C). Heracleous et al. [42] pointed out that most educational buildings in Cyprus are NV and consequently, cannot perform in a thermally comfortable manner (neither in the present or future climatic conditions). Similar conclusions were reported by Dhalluin et al. [83], who highlighted that NV strategies could be problematic in hot and humid climates due to the risk of overheating in NV buildings. Previous studies also investigated the use of fans in NV classrooms. Bhandari et al. [35] analysed the influence of ceiling fan-induced non-uniform thermal environment on thermal comfort in NV lecture halls in Tamil Nadu. They found a significant difference in thermal comfort conditions between different zones. Students seated farther from the fan experienced less air velocity, which impacted in thermal sensation, air velocity sensation, sweating sensation, air velocity preference, thermal comfort vote and productivity. Due to higher air velocity of fan, a slightly higher neutral temperature of 29.7 °C was determined in comparison with other studies conducted in similar climate zone. Other studies carried out in warm humid climate (Kumasi, Ghana) Koranteng et al. [84] concluded that opening windows and the use of fans could keep students comfortable.

These aspects are also relevant in NV offices buildings, as the intensity of fatigue and headache have been pointed out as one of the consequences of air temperature over the comfortable zone [85]. Kyritsi and Michael [86] carried out field measurement in office buildings, in the dry climate of Cyprus, and found that NV can have a positive contribution to the passive cooling of these buildings. He et al. [87] highlighted that office occupants in China place a great value on natural ventilation, especially in summer and winter.

It is also remarkable that in buildings where ventilation through doors and windows is the main way to control thermal comfort, occupant adaptation is particularly relevant. A considerable diversity in occupants' adaptation measures in NV buildings have been analysed in numerous studies. Indraganti et al. [73-76] found that occupants of apartments in Hyderabad (India) adapted through clothing and metabolism as the temperature increased in summer (although clothing adjustments in women were limited due to socially and culturally practices). At high temperature, occupants used fans and other electrical controls for air movement and windows/doors/curtains were closed to contain heat gain, glare and hot breezes. Kumar et al. [59] concluded that the proportion of 'open window' has a growing trend with an increase in indoor and outdoor air temperature values up to 34 °C. Dhaka et al. [67] reported similar conclusions in NV buildings in the region of India. In China, Lai et al. [88,89] analysed occupants' behaviour in residential buildings and the authors also concluded that NV duration increased as the climate became warmer. The study on thermal comfort in a family condominium conducted by KC et al. [90] in Shinagawa (Japan) found that opening the windows increased the thermal comfort of the occupants even though the indoor air temperature was quite high. Other adaptive measures were used, such as changing the insulation of clothing or using fans to adjust the thermal environment. Similar conclusions were found by Rijal [68] in Japanese houses during the hot and humid season. These adaptive behaviours in response to warm and hot conditions were also employed by households living in NV houses in hot humid climates in Australia, including adjusting the arrangement of clothing, operation of the dwelling and use of external measures (e.g. lowering body temperature by swimming), with an almost continuous natural ventilation strategy [91]. In contrast, previous studies identified barriers to use NV and adaptive measure in residential buildings in India [36,73]. Among the identified causes, it can be noted the lack of privacy when the windows opened into the public realm [73] or the heat outside [36] in dwellings.

Other studies have looked at the particular case of students' adaptive behaviour in NV educational buildings as thermal perception of children and teenagers can vary significantly from that of the adults [92]. Jindal [53] pointed out that students in this age range demonstrated considerable adaptability to indoor temperature variations by taking adaptive measures such as switching fans on/off or opening windows. These both measures were also identified by Khambadkone et al. [38], along with the wearing lighter clothing, partly closing curtain blinds and drinking cold water, as the main adaptive measures of students in NV classrooms in the warm-to-moderate climate of India. Aparicio-Ruiz et al. [93] observed that the preference towards opening windows and doors as adaptive strategy decreased along with the outdoor temperature in Spanish primary classrooms during summer season. Talukdar et al. [49] evaluated student thermal comfort and adaptation measures in NV university classrooms with ceiling fans in Mymensingh (Bangladesh) and concluded that high air velocity minimized the discomfort due to high humidity. These findings highlight the importance of occupant adaptation in NV buildings in warm climatic zones.

4.1.2. Strategies in MM buildings

Concerning MM buildings, they operate in both NV and AC. The measures of adaptive behaviour found in NV buildings are also employed by occupants in MM buildings. Additionally, occupants in MM buildings can also switch on and off the air-conditioning systems to control the indoor thermal environment [54,94]. The adaptive comfort behaviours in MM residential buildings in Tianjin (China) by Hou, J. [95] and they found that the most common adaptive strategies adopted by occupants to reach thermal comfort was opening windows and doors. Ke Zhong et al. [96] pointed out that occupants usually make full

use of the ventilating facilities or low energy-consumption methods to control and adapt the indoor thermal environment in NV and AC equipped rooms of residential buildings. Chen et al. [97] evaluated thermal comfort in a MM school building in Taiwan and also found that the most commonly used systems were open the windows and run the fans (64 %), which was followed by opening the windows alone (25 %) and the joint use of the fans and AC (11 %).

Regarding MM offices, Sun et al. [94] found that NV and cooling equipment (fans and AC) were commonly used in summer, indicating a MM adaptive behaviour. Indraganti et al. [70] found a low satisfaction in office buildings in Chennai and Hyderabad (India) when these buildings operated in NV mode. They observed that 80% of air velocity was around of 0.0–0.2 m/s, a range much lower than the allowable limits set in ASHRAE-55. This resulted in more than half the occupants demanded increased air speeds. Among the causes limiting the opening of windows, the authors identified the following: mosquitoes, noise, vandalism, jamming, user lethargy etc.

It should be remarked that the ventilation system influences the occupants' thermal perception. The comparison of several types of ventilation systems showed that the subjective thermal response of NV mode was significant different from that of AC mode. Lau et al. [58] conducted a study in educational buildings with AC, MM and NV systems in Singapore. A wider range of temperature acceptance was found in MM and NV spaces than in AC spaces. Despite this, the respondents of AC spaces had a cooler thermal sensation than MM and NV spaces, compared to NV spaces where the users presented a warmer sensation. This suggests that the occupants of these spaces have a higher tolerance to indoor thermal conditions. It allows the implementation of more adaptive measures in these spaces. Similar results were found in educational buildings in China [98], in Mexico [57] and in Australia [99,100].

In the case of offices, the comparative analysis among occupants' thermal perception led to locations around the world. Mustapa et al. [65] found that in FR-mode office buildings in Japan, many of the occupants would prefer slightly cooler conditions, while 50% of the occupants did not want to change the thermal environment in AC mode. The users of FR mode adopted an adaptive behaviour, such as taking drinks, switching on fans and opening doors and windows. Wu et al. [54] concluded that office occupants could benefit from split air-conditioned, since they could freely adjust indoor temperature and control windows, compared to the AC and NV buildings. This issue is also evident in the results reported by Rup et al. [101]. Lower prevalence of 'cold' discomfort response among the occupants of MM buildings was demonstrated, comparing those offices with centralized HVAC and under the equivalent thermal conditions in Brazil. In contrast, De Vecchi et al. [102] observed non-significant difference in terms of thermal acceptability between MM and fully-air-conditioned office buildings under humid subtropical conditions.

4.1.3. Passive cooling techniques

Numerous studies explored the use of other passive cooling techniques to improve the indoor thermal environment. For example, several studies analysed the used of night ventilation to control the indoor temperature in buildings [42,86,103–108]. Heracleous et al. [42] concluded that the night ventilation strategy in Cyprus has a positive contribution to the cooling effect of indoor spaces during the hot summer period [105]. Similar results were found in office buildings [86] and vernacular architecture [106,107] in this region. Kubota and Chyee [104] found that night ventilation is better than daytime ventilation in modern houses in Malaysia, while full-day ventilation and no ventilation were optimal in terms of air temperature reductions during the day and night. However, additional measures are necessary to improve diurnal thermal conditions in the room such as dehumidification (to reduce indoor humidity) or increasing the indoor air speed (using fans).

Other passive techniques, such as thermal stack flue, cross ventilation and water wall are also analysed in the research literature. Moosavi et al. [109] investigated the use these techniques in order to minimize the overheating inside an atrium lobby of office building in Putrajaya, Malaysia. From the comparison, they conclude that cross ventilation and water wall reduce the temperature inside the atrium when the stack flue has its highest performance. Nevertheless, the effect of stack flue is more effective on the reduction of the humidity. Elmualim [110] evaluated the use of wind catcher (tower natural ventilation system) in a seminar room in the University of Reading in the UK. This system, in conjunction with openable windows, can achieve a higher night and daytime cooling effect in the particular area where the system was installed. Furthermore, occupants expressed their overall satisfaction with the system and comfort conditions. Psomas et al. [111] analysed automated roof window control system to address overheating on renovated houses (Cfb climate) and found that automated window system reduces overheating risk in houses of temperate climates. Nevertheless, Ortiz et al. [112] analysed passive measures for the energy refurbishment of residential buildings in Catalonia (Spain) and pointed out that the behaviour of the passive measures is not always possible during the warm season, especially if the hours of overheating are analysed. Nevertheless, buildings may be constrained to operate naturally ventilated and cause thermal dissatisfaction for occupants

4.1.4. Modelling and assessment

It should be also highlighted the application of the several thermal assessment methods in warm-hot climates. Significant differences were detected between the assessment obtained from the static models and the adaptive models in NV buildings. Several studies pointed out that PMV method showed lower accuracy and occupants' thermal perception is often overestimated or underestimated [38,46,53,67,70,113–116]. Rangaswamy and Ramamurthy [113] analysed different thermal comfort indexes (i.e. PMV, e-PMV, a-PMV, Tropical Summer Index -TSI-, standard effective temperature -SET-, predicted thermal sensation -PTS-, ASHRAE-adaptive, and Indian model for adaptive comfort -IMAC-) in NV apartment buildings in Chennai (Hot–Humid Climate, India). Their results showed that PMV, SET and IMAC had lower accuracy compared to the other indices applied in this climate and building type, while the TSI is consistent in accurately predicting occupants' actual vote. The results reported by Wagner et al. [117] from a field study on thermal comfort in a NV office building in Karlsruhe (Germany) showed that models with a fixed limit o indoor temperature predict the thermal sensation and thermal comfort of occupants worse than adaptive comfort models, if periods with transient indoor (and outdoor)

climate conditions are considered. Lau et al. [58] conducted a study in educational buildings with AC, MM and NV systems in Singapore. They determined that the percentage of responses from respondents in MM and NV spaces (within the acceptable region according to the adaptive model) were consistently higher than those derived to the PMV/PPD model for all respondents. Similar conclusions were obtained by numerous researches in other climate zones [34,43,52,70,100].

In addition, previous studies concluded that the adaptive model provides a more accurate assessment than static models such as the PMV [34,39,54,55,95,118] in MM buildings. This is due to the possibility of using adaptive measures by the occupants, as they can switch between AC mode (switching on/off the air conditioning system) or NV mode (opening/closing doors and windows). The acceptable region of thermal comfort is wider in this type of buildings [66]. Deuble and de Dear [34] pointed out that adaptive comfort model was applicable to the MM building, especially during times of NV. One of the challenges of creating NV or MM buildings used by many occupants is not only to provide and ensure adaptive options, but also to foster a social context that encourages successful communication and implementation of adaptive adjustments among the occupants [119].

In summary, the models defined in ASHRAE-55 and EN 16798-1 were the most widely applied. Nevertheless, they resulted inaccurate in some climatic regions for assessing the thermal comfort of building occupants. To solve this drawback, several studies have proposed new adaptive models more suited to the climate of these regions. For example, Daghigh et al. [116] obtained that the neutrality temperatures in office rooms at University Putra Malaysia for both scenarios (NV and AC) were higher than those given in the ASHRAE Standard 55-1992 and suggested a wider thermal comfort band for Malaysians. Zhai et al. [52] analysed the use of adaptive control and its effect on human comfort in a NV office in California. They concluded that the adaptive model is generally effective in predicting thermal comfort, but its comfort zone could be wider for occupants in mild climates like in north California. Aparicio-Ruiz et al. [93] observed a widening in the thermal comfort range for children compared with EN 16798-1 and the ASHRAE-55 Standard in Spain. Bassoud et al. [44] discussed the thermal behaviour of NV buildings constructed in adobe in hot arid Sahara regions during summer and proposed a comfort model for this zone which is within the acceptability range of the ASHRAE model with a difference of order 4 °C, and an 8% difference between the slopes of the comfort temperature equations. Munonye and Ji [48] concluded that children schools are less sensitive to temperature changes than adults in NV primary schools of warm and humid environment in Nigeriaand that the comfort ranges. From this study, the upper limit of ASHRAE-55 comfortable temperature was extended. López-Pérez et al. [57] evaluated an adaptive thermal comfort field study in educational buildings in Tuxtla Gutiérrez (México) during warm season. The authors reported that respondents expressed thermal satisfaction at comfort temperatures above current international standards. Rawal et al. [39] also proposed an adaptive model for MM and NV residences in Indian climate.

4.1.5. Future perspectives

Finally, although advances in knowledge have been made so far, further studies are needed to explore the following aspects:

- Field thermal comfort studies in a wider variety of different climates. The literature review highlighted that many studies have been carried out in the regions of China, India and Australia. However, other warm climate regions, such as the Mediterranean basin and South America, are underrepresented in terms of the number of publications. Future field studies should be conducted in these areas to better understand the influence of climate on thermal perception and occupants' behaviour, as well as to generate a regional database, complete the ASHRAE global thermal database II and help to develop regional thermal comfort models applicable to NV and MM buildings in these locations.
- Analysis of different occupant activity types. Main studies focused on the analysis of thermal comfort in sedentary
 or near-sedentary physical activities (i.e. office work, teaching activities, domestic activities, etc.) Nevertheless,
 human thermal perception is influenced by metabolic rate, which in turn depends on the type of physical activity
 performed. It is necessary to extend studies to other types of activities, considering different metabolic rates and
 assessing the impact of building performance on the thermal comfort of users in these scenarios.
- Elderly building occupants. More research is needed to understand the impact of thermal conditions in NV or MM buildings on older occupants. Society in developed countries is facing demographic change due to an ageing population. In Europe, 29.1% of the population is expected to be over 65 by 2080 [120]. Given that the thermal perception in this age range is different from that of an adolescent or adult, and that global warming may lead to more heat waves in the coming years, analysing this segment of the population is essential because of their special vulnerability to the risk of overheating.
- Further analysis of adaptive measures. Numerous studies reported that adaptive measures, such as the use of electrical appliances like fans, can improve the thermal satisfaction of building occupants [35,121]. However, these measures may also cause local discomfort (due to the relative position of the occupant inside the room and that the air velocity may be higher/lower) and influence other domains of indoor conditions (e.g. noise). In this respect, further research is required to study the influence of this equipment in the overall occupants' satisfaction.
- Previous studies suggest that subjects' physiological adaptation may not be influenced by short-term indoor thermal history. However, further studies are required to confirm that there is no effect [98]. This aspect should be analysed to identify its impact in the behavioural adaptation aspect (adjusting clothing and using fans, doors, windows and AC).
- Different types of occupants can be in a building simultaneously. Research focused on the study of adaptive measures among office workers or occupants of residential buildings has been extensively evaluated. Nevertheless, other scenarios where different types of occupants come together (i.e. students and professors, workers and

users/costumers, etc.) should be evaluated and adaptive measures for each category and type of activity should be assessed.

- Analysis of the passive cooling strategies' usage in contemporary buildings. Previous studies highlighted that some buildings cannot perform thermally comfortably under future climate conditions [42] and a risk of overheating might be expected in future years [81]. It needs future research efforts to analyse passive cooling strategies to effectively limit the overheating risk in contemporary buildings.

4.2. Implications of ventilation in the energy performance of buildings

4.2.1. The impact of climate change in the use of NV

The use of NV to improve the energy efficiency of buildings has focused much of the research in recent years. The possibility of using NV is simple: if the outside temperature is within the user's thermal comfort conditions, the interior can be ventilated. This makes possible to achieve reductions in cooling energy demand. Although the premise is simple, there are many factors that affect the possibility of applying NV effectively, such as the urban environment [122] or the shape of the building [123,124].

NV is understood as an effective measure to reduce energy consumption in the warm months. In this sense, its use during the summer stands out [125]. This aspect may be an advantage in the context of climate change, since summers are expected to be longer. However, a significant increase in outside temperature is also estimated, which could limit the use of this strategy. In relation to this, the studies carried out by Bienvenido-Huertas et al. [126,127] have reflected the expected variation with the use of NV. Figure 9 shows the trends of change in the use of NV in Europe with RCP scenarios. As seen, the northern areas will increase the use of NV throughout the year, while the southern areas will see a decrease in the possibility of use.



Figure 9. Maps of Europe and the Mediterranean area with the yearly use of NV (Source: Elaborated by the authors). Data and methodology used according to the study by Bienvenido-Huertas et al. [128,129].

In any case, climate change will mean a variation in the possibility of using NV to save cooling energy consumption. This aspect has already been reported by several studies: (i) Kaudeer et al. [130] determined that climate change will increase the hours of thermal discomfort in naturally ventilated buildings in Mauritius by up to 15%; (ii) Rysanek et al. [131] evaluated the impact of climate change on naturally ventilated office buildings in Vancouver. The analysis was carried out for the summer months using the adaptive thermal comfort model of ASHRAE 55-2017. The authors found that the RCP 8.5 scenario may increase by up to 33 days per year in 2050 and up to 55 days per year in 2080; and (iii) Pajek et al. [132] determined that the most effective measures for energy saving in the residential environment in Montenegro were roof insulation, shading and NV. Thus, the use of NV has an expected variability in the future, which implies increases or decreases

in its effectiveness as an energy measure. The results based on the current scenario have shown interesting energy savings, but that are far from zero energy consumption. Li et al. [133] determined that night ventilation in Shenyang University buildings (China) allowed cooling energy savings of up to 45% in summer. Sun et al. [134] focused their efforts on finding effective passive measures for disadvantaged neighbourhoods in California. The results showed how NV allowed to achieve reductions of up to 26% in hours of thermal discomfort. Thus, these studies did not show 100% effectiveness of NV. For this purpose, it is necessary to combine NV with other measures. In this sense, Sun et al. [134] highlighted that in their study the use of NV with other passive measures and air-conditioning to achieve 100% of the hours of thermal comfort.

4.2.2. Combination of NV with other saving strategies

Given this circumstance, much of the efforts of researches have focused on the optimal combination of ventilation strategies with other measures [135]. The analysis of NV energy savings with other measures is very varied (Figure 10). Analysis with variables such as: building shape [136–138], shadows [139], Windows to Wall Ratio (WWR) [136–138], orientation [136–138], cold roof and walls [136–138], thermal transmittance [140], solar heat gain [141], courtyards [142,143], and Thermally Activated Building Structures (TABS) [144].



Figure 10. Combinations of NV with other measures analysed in the scientific literature. The number of studies associated with each combination is also indicated (Source: Elaborated by the authors)

In these studies, it is possible to highlight those that have been oriented in climatic environments of desert areas, traditionally hot. The construction and ventilation techniques of these countries can be followed to propose measures that are useful throughout the world. In this sense, Zoure and Genovese [136,137] and Elshafei et al. [138] analysed the effectiveness of NV with other factors (building shape, glazing, shading and WWR) in buildings with an arid climate, such as Burkina Faso or Egypt. The results showed that the use of NV together with north-south orientations and WWR of 30% allowed to achieve reductions in yearly energy demand of up to 15%. The results also reflected the need to avoid large window dimensions to reduce solar radiation. A simpler analysis than the previous ones was carried out by Elhassan [141] in an arid climate. In his research, Elhassan determined that the unique use of NV allowed energy savings of up to 70%. Likewise, Elshafei et al. [145] observed that modifications in window design (window location and size, as well as the

incorporation of windows) in residential buildings in Egypt allow decreases in indoor air temperature which implies a minor air-conditioning consumption. Khalid et al. [139] analysed the possibilities of efficient cooling of buildings located in Pakistan. Cross natural ventilation led to achieve reductions in energy consumption, although its combination with External Thermal Insulation Composite Systems (ETICS), shading blinds and low emissivity glass allowed to achieve an optimal reduction in energy consumption.

Likewise, the use of NV together with front green wall can also be an effective measure in climates such as Pakistan [146], with reductions in cooling energy consumption of up to 27%. Finally, El-Bichri et al. [147] evaluated how natural night ventilation and shading techniques reduced the energy demand up to 51% for cooling rammed earth houses in arid climates. Along this line, Albatayneh [148] performed a parametric analysis of building design optimization. Through the sensitivity analysis with the results obtained for Jordan, 2 clusters of measures were detected based on their importance. The most important variables were NV together with WWR, shadows, infiltrations, and type of glazing. These variables had more relevance with respect to others, such as orientation or the type of shading element. Akhozheya et al. [149] evaluated the application of high thermal mass measurements and night ventilation in an educational building in Minneapolis (United States). Up to 3°C could be decreased of the maximum temperature values in the classroom and with delays of 4 hours with respect to the designs of average thermal mass.

The combination of bioclimatic designs with NV was also relevant. Bevilacqua et al. [150] analysed the possibility of applying NV and Trombe walls in buildings located in the Mediterranean climate. The results obtained decreases in the cooling energy demand, in such a way that the systems were activated during less than 10% of summer hours. The effect of inner courtyards can also have a significant impact on energy savings with NV. Gunasagaran et al. [142] determined that the combination of semi-enclosed courtyards with shades could reduce the use of air-conditioning systems in warm climates. However, the effectiveness of NV may have rebound effects on other types of energy consumption. Likewise, Sun et al. [143] determined 7 courtyard design guidelines to achieve an adequate decrease in cooling energy demand. These guidelines included everything from courtyard height to stepped designs. In any case, the analysis was limited to the climatic and architectural characteristics of Vietnam. Likewise, other case studies have been evaluated for special buildings. For example, in high-rise office buildings, the large number of windows can increase cooling energy demand from solar gains in summer. The use of NV can compensate part of this energy demand, although the support with other solutions is necessary [151]. In this sense, transom window is also an interesting measure to improve the energy performance of this type of building. Liu et al. [152] concluded that the mentioned measure presented cooling energy savings of up to 22.7% in high-rise buildings. Hence, NV in combination with passive cooling measures can achieve significant decreases in energy consumption in the warm months. Despite this, passive and NV measures cannot prevent thermal discomfort in many of the studies in the scientific literature [134,146]. To avoid this, it will be necessary to use MM to be able to compensate for the hours of high temperatures [153].

The potential of MM for energy savings is clear. Indeed, several studies have highlighted the following aspects. Cook et al. [154] analysed the impact of NV and MM on the air-conditioning hours in Indian residential buildings. The use of MM led to reductions of up to 55% in energy consumption. Ran et al. [155] managed to reduce the number of air-conditioning hours in Chinese residential buildings by 50% with the use of MM. Sánchez-García et al. [156] evaluated the use of MM in office buildings in southern Spain, obtaining a decrease in cooling energy demand of up to 74%. Bienvenido-Huertas et al. [125] evaluated the use of MM in social dwellings in coastal areas of southern Spain. The use of MM based on EN 16798-1:2019 made it possible to achieve reductions of up to 100% in cooling energy consumption, as well as the risk of energy poverty in families [160]. Thus, MM could guarantee 100% hours of thermal comfort in warm months with moderate energy consumption. The greater or lesser effectiveness in energy saving depends on the operational pattern considered for NV. Approaches based on adaptive thermal comfort models such as ASHRAE 55-2017 and EN 16798-1:2019 have shown great effectiveness in previous studies [125,156]. However, these approaches can have very wide ranges of temperatures and limitations in certain buildings, such as elderly care centres [157].

Likewise, the use of a more efficient MM has also been assessed with modern equipment and passive measures similar to those analysed in the NV studies. Shakya et al. [158] evaluated the potential of using MM with radiant ceiling cooling systems and dehumidifiers in hot climates such as Singapore. Up to 77% less of energy consumption was obtained in comparison with classic cooling systems. MM based on the combination of NV wind collectors and evaporative cooling is also an opportunity in hot climates. It is expected that the cooling system will reduce its consumption by 52% in school buildings in Kuwait [159]. The use of Earth-to-air heat exchanger can also be an alternative in summers of warm climates, with inlet air temperature reductions of up to 9.9 °C [160]. MM can be improved by incorporating ceiling or manual fans [161] (because air speed increases convection around users [162]). These systems could achieve a greater reduction in cooling energy consumption in hot seasons, as reported below. Bamdad et al. [163] found that the use of MM with ceiling fans led to savings in Australian buildings of up to 23% today and 15% in the future. André et al. [164] analysed the use of MM with desk fans in office buildings in Brazil. The results showed that energy consumption could be reduced by up to 40%. Bienvenido-Huertas et al. [165] evaluated the application of the tolerances of the upper limit that admits the standards of adaptive thermal comfort with the increase of the speed of the air in interiors. Up to 600 hours less of air-conditioning could be achieved by fans that generate air speeds greater than 0.6 m/s. Furthermore, MM can be optimized for future scenarios by using modern installations. In this sense, Martins and Bourne-Webb [166] determined that the use of NV together with ground-source heat pump system can be an effective measure to reduce the cooling energy consumption of office buildings in Spain in the future.

It is also convenient to highlight the great potential that MM (or NV) has in the event of possible power outages or overheating, especially in heat waves [167]. Borghero et al. [168] stated that, although the use of active systems would allow more hours of thermal comfort to be achieved in future heat waves, the high peaks of energy consumption in these periods could imply to cuts in the electricity supply. The use of NV with the adaptive approaches of EN 16798-1:2019 could lead to

reductions in energy consumption of up to 15% in future heat waves. In any case, overheating can also be avoided with NV or MM [169]. For this purpose, it uses the experience of countries with desert climates (such as western sub-Saharan Africa [170]) in which NV together with other passive measures decrease the risk of overheating. Thus, NV is essential to improve the performance of passive measures, since even without NV passive measures can have negative effects. In this sense, Bo et al. [140] determined that the insulation of rooms without NV could lead to an increase in superheated hours of up to 22.5% compared to rooms without insulation. Likewise, Baba et al. [171] determined that high energy efficiency buildings have risks of overheating if they do not have adequate ventilation. Thus, building energy improvement strategies should consider ventilation to avoid the risks of overheating. Despite the aforementioned aspects, there is the effect of incorrect applications of MM. Du et al. [172] evaluated 10 households in Chongqing (China) for one year. The results showed that up to 68% of the users left small areas of the windows open when they had the air conditioning system activated. This caused an increase in the cooling energy demand.

4.2.3. Future perspectives

As seen, there are a series of knowledge gaps about NV and MM that have been detected and that should be analysed in future works of the theme:

- Climate and effectiveness of NV and MM. Many studies report the climate as one of the main limitations. The results obtained could vary in other climates [136–138]. Likewise, the characteristics of the built environment may also vary. It is necessary to deepen the differences between energy saving solutions. For this, interregional studies should be carried out where the effectiveness of the holistic strategies considered (NV or MM with other measures) is known in various climates at the same time.
- Holistic analysis of NV and MM with other energy saving measures. In the state-of-the-art, many studies have addressed the use of NV (or MM) with other passive measures to reduce energy consumption and improve the thermal comfort of users. The existing results in the literature have shown that the combination of measures is useful to achieve significant reductions in energy consumption. However, the combinations used in the studies do not usually consider all possible measures. To eliminate this knowledge gap, the authors should perform holistic analyses where the different variables indicated in Figure 4 are combined.
- Influence of the use of NV or MM. Few studies have addressed the joint influence of various factors on the implementation of NV or MM. These factors may be different from the indoor temperature. In this sense, the work carried out by Belias and Licina [173] reflected how outdoor air pollution (mainly due to PM2.5 and PM10) limited the energy savings obtained with NV. Thus, the location of the building in the urban environment is essential to assess the effectiveness of ventilation, since areas with high contamination will have a low potential for cooling savings. In any case, the scope of the study was limited without considering other potential environmental factors that may affect energy-saving potential, such as outside noise [174,175]. Future work should address the limitations associated with the environment of cities in the use of NV.
- Energy saving and thermal comfort with night ventilation. Several studies report the advantages of night ventilation for energy savings and thermal comfort hours [176,177]. They achieve this both in office buildings and residential buildings. However, these studies do not consider the effect that nocturnal ventilation has on sleep quality [178]. The consideration that the thermal comfort conditions during sleep hours are the same as sedentary activities can generate errors in the assessment of the thermal quality of the indoor environment. Thus, users may not endure the use of NV at night. This would mean that the current results could overestimate the potential for energy savings in residential buildings. Future studies should address the existing limitation in energy savings with night ventilation in residential buildings.
- User participation and energy saving. The effectiveness of NV in achieving energy savings depends to a large extent on the participation of the occupants. Most of the studies tend to consider that NV is regulated by controlling the environmental temperature (indoor or outdoor) [125]. However, this may be unrealistic if users do not have the means to perform a check. Thus, future work should address the possible limitations associated with the effective operation of windows to achieve energy savings. Likewise, knowledge of other measures that the user can adopt to reduce the energy demand for cooling should be expanded, such as manual fans or the use of light clothing.
- Application in different case studies. Most NV studies have focused efforts on residential buildings [168,179], office buildings [136,137] or educational buildings [153]. Figure 11 summarizes the studies published by type of building case. As observed, there is a significant difference with other types of buildings. It is necessary to extend the analysis to other case studies where NV may have more limitations. An example may be the elderly care centres. The main limitation is the adaptability that older people have, especially in hot seasons. Faced with this circumstance, many studies opt for other type of measures, such as the energy improvement of the building and the moderate use of air conditioning [157].



Figure 11. NV or MM studies by type of building (Source: Elaborated by the authors).

4.3. Indoor air quality and ventilation in warm climate zones

The quality of the air inside and around buildings and structures is refers as Indoor Air Quality (IAQ). The application of minimum ventilation rates and other measures to avoid poor IAQ is essential to minimize adverse health effects [180]. Several air pollutants, such as gases (e.g. carbon dioxide, radon, or volatile organic compounds), particles (e.g. PM10 or PM2.5) and bioaerosol (e.g. mold or virus) may deteriorate the IAQ and affect occupants' health and well-being. Figure 12 summarises the published studies by pollutant type. The Environmental Protection Agency noted that inadequate ventilation can result in indoor pollutants becoming more concentrated by not letting in enough outdoor air to reduce emissions from indoor sources and by not removing pollutants from the area. Therefore, maintaining safe concentrations of air pollutants for building occupants requires sufficient air exchange through ventilation. NV can be used to supply fresh outdoor air to indoor spaces and to reduce indoor generated pollutants.



Figure 12. Number of documents by type of air pollutant (Source: Elaborated by the authors).

4.3.1. Carbon dioxide and carbon monoxide

Great attention was paid in scientific literature to carbon dioxide (CO_2) concentration because, although it is not classified as a pollutant by the World Health Organization, high concentrations may cause negative effects in occupants and international standards used this parameter as an indicator of IAQ (ASHRAE, RHEVA). Several studies monitored CO_2 concentration in NV buildings during warm season. Shrestha et al. [181] found that indoor CO_2 concentrations were significantly lower than the acceptable limit in NV school buildings in the temperate climate of Nepal during summer season and that most students perceived the IAQ as acceptable under such conditions. Habil and Taneja [182] highlighted that the CO_2 concentration values were lower in summer than winter due to the lower ventilation rates observed in densely populated classrooms during cold season. A higher correlation between CO_2 and ventilation was obtained in summer season, which explains the decay of CO_2 levels when exchange takes place through open windows and when classrooms are unoccupied, showing the ability to building ventilation. Campano-Laborda et al. [183] stressed that the use of CO_2 as a standalone indicator of environmental quality may be insufficient and lead to situations of increased discomfort for occupants. Student' environmental sensation votes, perception and indoor-related symptoms (i.e. tiredness) have a direct influence in windows' operation during lessons in Southern Europe. In fact, effective controlled ventilation systems are required to guarantee a real renewal of the indoor environment and a supply of clean air in NV classrooms. Similar results were reported in studies conducted in residential buildings [184], offices buildings [185] and kindergartens [186]. In contrast, the results reported by Yang et al. [187] from a field study conducted in Korean schools suggested that air exchange rate was too low in classrooms during summer season since the mean CO₂ concentration was above 1000 ppm.

In any case, NV supposes a challenge to stop outdoor pollutants from entering the building. Given window opening time increases as temperature rises in NV buildings, ventilation rate also increases in summer and indoor levels of air pollutants are strongly dependent on outdoor ambient levels [188,189]. Thus, numerous studies monitored other air pollutants generated both indoors and outdoors. A pollutant that can be generated both indoors and outdoors is carbon monoxide (CO). Indoors, the generation of CO is mainly due to the malfunctioning of gas installations such as a faulty gas furnace with a blocked exhaust pipe. In this context, a proper NV method together with optimal placement of the header in the household minimize the risk of CO poisoning [190]. However, this situation is extraordinary and previous studies pointed out that indoor CO concentration were lower than those of outdoors, suggesting that the mains CO source was outside the building (e.g. traffic) [187,191]. Chithra and Nagendra [191] found low concentrations of CO in school buildings, although there were higher concentration values during peak traffic hours. Abdel-Salam [188] monitored CO in residential buildings in Alexandria (Egypt) and found a season variation in CO levels which could be due to different ventilation practice in winter and summer.

4.3.2. Particulate matter and volatile organic compounds

The particulate matter (PM) is a complex mixture of solid and/or liquid particles suspended in air that can be found indoors and outdoors [192]. This parameter is also commonly monitored in NV buildings and, depending on the location of the buildings, the outdoor concentration may be higher or lower than the indoor concentration. Habil and Taneja [182] analysed the PM 10, PM 2.5 and PM 1.0 levels inside and outside NV school classrooms, and [188] the mean indoor concentrations lowered from 3 to 2 times in summer compared to winter. Chithra and Nagendra [193] carried out a measurement campaign in a NV school building located near an urban roadway in Chennai (India). The authors found a strong seasonal influence with minimum PM values during summer and maximum values during winter and monsoon seasons. The indoor PM 10 and PM 2.5 levels were often exceeding the National Ambient Air Quality Standards. Similar results were found by Mentese et al. [194] for PM2.5 in different type of buildings (house, office, kindergarten and primary school). In contrast, other studies found opposite trends in the ratio of PM levels in winter and summer. Oroji et al. [195] evaluated the PM concentration in NV classrooms at Shahid Beheshti University (Tehran) and found that the lowest concentration occurred in spring and winter, while the highest concentration in summer. The trend of variation of concentration inside the classroom was similar to the outside due to the NV in spring and summer seasons. Sahu et al. [196] assessed spatial and seasonal variations of IAQ among different indoor microenvironments of a technical university in India and concluded that PM level with open windows (NV) showed higher seasonal variation compared to AC mode. Furthermore, Belias and Licina [197] performed simulations to assess the use of NV in residential buildings in 26 European cities and pointed out that both PM2.5 and PM10 significantly restricts the use of NV in all types of locations evaluated. In order to minimize PM pollutants concentration, previous study analysed the effect of air purifiers on the reduction of indoor PM2.5 in NV buildings and found that air purifiers are effective for indoor PM2.5 control. Li et al. [198] concluded that air purifiers can reduce population health damage by 43.47% to 86.46%.

As for Volatile Organic Compound (VOC), Mentese et al. [194] found that VOC levels were higher in winter than summer. They observed that, among the VOCs, toluene was the most predominant, whereas elevated n-hexane levels were also observed in the kindergarten and the primary school, probably due to the frequent wet cleaning during school days. Stamp et al. [199] found that indoor generated VOC pollutants (d-limonene and alpha-pinene) in offices were reduced by up to half with increasing summer NV.

4.3.3. Biological pollutants

Regarding biological pollutants, its control is essential to ensure suitable IAQ and prevent the transmission of airborne viruses. Particularly, an adequate management of IAQ in healthcare units and hospitals has relevant impact on patient safety and occupational health. The microbiological loads of indoor air in healthcare units are of utmost importance since the airborne spread of pathogens is one of the main causes of nosocomial infections. Fonseca et al. [200] found an increase in microbiological in units care during summer caused by higher values of indoor air temperature and RH. Núñez and García [201] found that opening a window in a hospital for 2 h has little effect on the composition of bioaerosols in the room (any season). Mendes et al. [202] assessed elderly care centers and found low levels of bacteria concentration, but indoor levels were twofold higher than levels outdoors. NV can be a useful addition to mechanical ventilation systems in healthcare facilities for IAQ management. It is important to keep the indoor air relative humidity and CO₂ concentration at appropriate levels to lower the chance of infection through the air [200]. Moreover, due to the COVID-19 outbreak, recent research has analysed different NV strategies in relation to whether they provide sufficient ventilation rate to minimize airborne virus transmission in offices [203,204], educational buildings [205–207] and residential buildings [208,209].

4.3.4. Radon

Along this line, ventilation has also proven to be an effective measure for other pollutant gases such as radon. In the case of this gas pollutant, the combination of ventilation together with high temperatures favours low levels of indoor contamination in summer [210]. The results reported by Al-Azmi et al. [211] suggested that the lack of ventilation in the basements may result in higher concentrations of radon levels compared with other rooms on the ground and first floors of dwellings. NV reduces radon levels both by dilution and by decreasing basement depressurization and thus the radon entry rate.

4.3.5. Impact of ventilation strategies in IAQ

Within this context, the scientific literature shows that NV can be an effective measure to dilute indoor pollutants to levels that are not detrimental to the health and well-being of occupants. However, the opening behaviour of windows by occupants is a key factor affecting IAQ conditions, as it is de only way of air exchange between indoor and outdoor in NV buildings [212]. As a result, IAQ may be compromised on occasions when external conditions are not favourable (e.g. hot calm days) [213]. Therefore, the occupants' behaviour is a key factor in the analysis of NV and IAQ that has been analysed in previous studies. For example, the stimuli for the use of windows in offices in South Korea were analysed by Yun et al. [214]. They found large seasonal effects on window use patterns and drivers for the use of windows would differ with season. During summer, the use of a window is influenced by CO₂ concentration when the window is already open, but it is not related to thermal factors. In addition, the probability of a window being opened increased as the CO₂ concentration rise. This fact suggest that perceive IAO is a non-thermal stimulus affecting the applicability of windows. In contrast, if the window is closed in summer, the use of a window is explained by thermal stimuli while the CO₂ concentration level is irrelevant. The transition probability of a window state from closed to open rises with an increase in indoor temperature and RH but falls with an increase in outdoor temperature and RH. Yun et al. [214] inferred that, when the outdoor temperature is high, the occupants would realize that opening a window would increase indoor temperature from their past thermal experiences of hot summer. In this scenario, occupants would prioritize thermal satisfaction over IAQ, and may therefore be exposed to poor IAQ. This aspect was also highlighted by Deng T. et al. [212], who identified that the tree main drivers for occupants' window-opening behaviour in the temperate zone in China were: indoor CO₂ concentration, indoor temperature and outdoor PM10 concentration. The effect of unhealthy IAQ conditions in indoor spaces with AC mode was noted by Graudenz et al. [215], who highlighted that a group of individuals working in an office building under AC systems has a higher prevalence of nasal, sinusitis symptoms, naso-ocular, persistent cough and building relation of the symptoms, than individuals under NV.

Similar results were found by Lai et al. [88,89] in MM residential buildings, where the researchers found that as climate became warmer, natural ventilation increased and mechanical ventilation decreased. However, the open-window duration starts to decrease when an outdoor temperature is reached. Above this 'cut-off' temperature, occupants tend to use AC systems and close windows. Lai et al. [89] stated that the 'cut off' temperature ranged from 24 to 29 °C in different regions of China, indicating that building occupants in southern regions were more prone to use AC than those in northern warm climates. Du et al. [216] added that although the probability of opening windows increases with increasing indoor/outdoor temperatures, the monitored daily durations for opening windows reduced when the indoor temperature exceeded 30 °C. In fact, they observed that the ratio of duration of the day for windows opened is higher than 0.8 when the outdoor temperatures ranged between 15 and 25 °C, while this proportion decreases to <0.6 when the outdoor temperature is outside this range. During periods in which windows are closed and AC is on, indoor air may cause uncomfortable feelings like stuffy air and poor air quality in residents 'subjective perception since enhanced airtightness in rooms would increase the CO₂ concentration. These results showed the importance of understanding the occupants' window-opening behaviour. It is essential to analyse ventilation performance and design the required outdoor air supply rate in rooms for comfort and health. Nevertheless, previous studies have already pointed out that there is a disconnection between observed window opening practices and typical design principles, assuming adjustment to a given ventilation system or changing weather conditions [217].

It is also worth noting that, although cross ventilation is a more effective ventilation strategy than unilateral ventilation, there are spaces where this ventilation strategy cannot be applied (e.g. studios and one-bedroom flats). In these cases, proper design of balconies and openings can significantly improve indoor ventilation performance, and thus improve both IAQ and thermal comfort [218]. Furthermore, the use of mixed ventilation (i.e. fans + NV) in classrooms proved to provide higher ventilation rate values compared to the exclusive use of NV [219]. Several studies also assessed the stack ventilation effectiveness in term of IAQ [220–223]. The wind catcher systems are considered an efficient way channel fresh air into the indoor spaces and can be used to reach sufficient ventilation rates in public buildings [224]. Elmualim [110] evaluated the effectiveness of a wind catcher and found that the CO₂ concentration was within standards and rarely exceeded 500 ppm. Jones and Kirby [225] analysed the IAQ in UK school classrooms ventilated by natural ventilation wind catchers and concluded that the ventilator is capable of significantly improving ventilation rates as well as reducing CO₂ levels during summer months, especially when used in combination with open windows. Kolokotroni et al. [213] investigated the performance in use of ventilation stacks in classrooms of a school during the summer period and found that the measured classroom air exchange rates did not differ between rooms with stack and without stacks during one measured period. This fact reflects that occupants would adjust flow paths to increase ventilation in the classroom (by opening the windows fully or opening internal doors).

4.3.6. Futures perspectives

In this section has been identified several research gaps that need to be addressed in future research, such as:

- IAQ field studies in different locations. Several studies analysed the effect of NV on IAQ in different locations in the urban environment [182,193,194]. However, further research involving a wider variety of locations (urban environments, industrial, dense road traffic, etc.) is still needed to analyse the effect of NV or MM on occupants. The growth of urban areas and the densification of cities may lead to an increase in air pollution and therefore affect IAQ inside buildings. Hence, studies that evaluate and provide control strategies to avoid poor IAQ in NV and MM buildings in these scenarios are needed.
- Analysis of the potential of passive ventilation systems in different regions. Numerous scientific studies have highlighted the use of solar chimneys, wind towers and other passive ventilation systems [17]. The implementation

of these systems could contribute to improving IAQ inside buildings in hot regions. Future studies should be developed to assess their implementation in current and future scenarios. Furthermore, it is necessary to assess their impact not only on-air quality or thermal conditions, but the effect on indoor environmental quality as a whole should be assessed including other factors such as noise and lighting.

- Seasonal variations on IAQ have been scarcely analysed in the scientific literature [188,196]. Future research is needed to analyse this issue in detail in different warm locations. In fact, many studies are limited in demographic, geographic or temporal scope [195]. Wider research is required to analyse adverse health effects derived from IAQ conditions on occupants of NV and MM buildings.
- Impact of IAQ on MM Buildings. Numerous studies have evaluated the impact of IAQ in NV buildings, but few have focused this evaluation on MM buildings. The use of AC mode can lead to decreased ventilation rate and increased concentration of indoor pollutants [216,217]. In fact, studies found a higher prevalence of nasal symptoms in office buildings under AC system [215]. It is necessary to deepen the analysis and evaluation of IAQ in these spaces under different modes of operation.
- Analysis of the combination of air filters with solar chimneys and windcatchers. Although both systems have shown their potential application in buildings, no studies have been found that combine them with air filters [17].
- More studies on the relationship between thermal comfort and IAQ are needed to better understand human behaviour and NV strategies [202]. Few studies analysed the effect of thermal comfort and IAQ on human behaviour as a whole, but a holistic analysis that takes into account different variables such as indoor temperature and pollutant concentration is essential [88,89,216,217]. This approach will allow a better understanding of window-opening behaviour or mode switching between NV and MM.

4.4. Simulation tools for ventilation

Several computational simulation methods have been identified for NV studies in areas or regions with hot climates: (i) simulation based on CFD, (ii) Building Energy Simulation (BES) and (iii) combination of CFD with BES tools. These approaches were mainly executed to evaluate NV performance, energy/thermal building performance or a combination of both. Figure 13 shows a summary of the simulation process used in the reviewed studies.



Figure 13. Simulation process used in NV and MM buildings (Source: Elaborated by the authors).

4.4.1. Simulations based on CFD

CFD is a method used to evaluate indoor environments by analysing airflow and physical environmental parameter distribution (air velocity, pressure, temperature and particle concentration) in indoor spaces. CFD uses numerical analysis and turbulence models to solve problems that involve fluid flows. The principles of the Navier-Stokes equations are employed in these models [226]. Previous research that has carried out CFD simulations includes the use of software such as ANSYS Fluent [227-229], TRNSYS [230-232], OPENFOAM [233], PHOENICS [234-236], Design Builder [237,238], CONTAM [190,239], STAR-CCM+ [240], Solidworks [241], CFX [242], AirPak (Y. Hou et al., 2022), ANSYS ICEM CFD [244,245], Comsol [246], Gambit [247,248], FloVent [249]. CFD has been extensively applied for airflow analysis. For example, numerous studies used CFD to assess the configuration of courtyard space to enhance environmental conditions, since it is essential to mitigate outdoor climate fluctuation in buildings in hot-humid climate [227,228,250-254]. The reported results showed that the use of courtyard along with NV strategies can provide effective building heat dissipation. Similar studies computed CFD to assess NV mechanisms in vernacular buildings [233,243,255] and residential buildings [229,249]. Along this line, CFD has been also implemented to simulate ventilation patterns with different ventilation systems, such as windtower/windcatcher [234,244,256,257], solar chimney system [241], naturally ventilated roofs [235,248,258] or night ventilation [259]. Previous studies also employed CFD to simulate the combination of different passive cooling system: solar chimney and radiant cooling cavity [260], windcatcher, earth-to-air heat exchanger and direct evaporative cooling system [261], natural ventilation and passive wall system (evaporative cooling) [245] or solar chimney and windcatcher [262], as well as migration of pollutants inside buildings [190].

4.4.2. Simulations based on BES

Regarding BES tools, they are mainly developed for the calculation of the thermal loads of HVAC systems and building energy performance. BES systems usually adopt multi-zone approach based on airflow network model [226], and each room

is considered a separate zone [231,263–266]. The accuracy of the multi-zone modelling increases compared to unizone, although it increases model complexity and computation time. For this reason, previous research also includes other software such as EnergyPlus [267–269], Design-Builder [270], IESVE [271–273], IDA-ICE [274,275], BSim [276], DOE-2 [277], TRNSYS [231,278], ESP-r [279], Modelica [209], WinEtana [280], TAS software [281–284], CAPSOL [285], ECOTECT [286–288], H-AIR [289], COMETh [290], CitySim [291], DIAL+ [292,293], Dest [294,295], EQUEST [296]. For those studies that considered a single-zone model to represent the building system for being simpler and faster, such as type building [267,268,273,297–300], it was extrapolated that the heat transfer between different spaces was not adequate.

Concerning the simulation of NV buildings' performance with different characteristics and configurations, El-Bichri et al. [278] evaluated the impact of different construction materials on building's thermal behaviour and indoor thermal comfort in a hot and semi-arid climate. Ben-Alon and Rempel [297] investigated different earthen and conventional wall assemblies to provide adaptive thermal comfort in different climates (US), with and without passive heating and cooling systems. Other studies also investigated the use of phase change materials in conjunction with the use of NV strategies to improve the energy efficiency of buildings [263,267,270,301]. The literature review also noted the applicability of BES for the assessment and simulation of passive cooling strategies in different types of buildings, such as: natural ventilation, wall absorptance, shading effect of the semi-outdoor space as perimeter buffers (e.g. balconies), solar heat gain coefficient of windows, wall U-value, WWR, building orientation or passive direct evaporative cooling systems [268,269,298]. In addition, building retrofit measures and their impact on energy savings and thermal comfort was studied. BES allowed the life-cycle cost of different building solutions to be assessed, enabling the selection of optimal retrofitting strategies based on longterm costs and benefits [264,299,302,303]. Another approach investigated in previous studies is the analysis of human behaviour in NV buildings. For example, Schünemann et al. [274] demonstrated how the window ventilation behaviour affects the heat resilience in multi-residential buildings, or the study conducted by Scheuring and Weller [300] where they analysed the impact of different control strategies on the energy efficiency of an operable louver window. Other researches focused on the use of MM in residential buildings [266,304] or office buildings [265,305].

4.4.3. Coupled simulations between BES and CFD

The use of coupled simulations between BES and CFD can provide a more accurate prediction in NV buildings. Indeed, there are numerous combinations of both types of programmes in the scientific literature. For example, the BES software ESP-r and the CFD software Fluent can make possible to estimate the impact of various ventilation strategies and façade designs on the indoor thermal environment of NV residential buildings [279]. This process was used to assess various façade designs according to the orientation, WWR and shading device dimensions. The combination of both software provided a quicker and more accurate method of assessing the NV performance of the building and detailed thermal environmental information. Liping and Hien [279] developed a coupling program that fed the indoor CFD simulation with the boundary conditions previously calculated with ESP-r. The DesignBuilder's CFD and EnergyPlus programmes have also been used together by Prajongsan and Sharlples [306] to assess thermal comfort and energy saving in high-rise residential buildings using ventilation shafts in Bangkok. The parameters included in the built model were the location, building' components, opening sizes and operation schedule, internal heat-gain and weather data. The boundary conditions were first calculate using EnergyPlus and subsequently used for the CFD simulation. The same software combination has been applied for other purpose in NV buildings, such us to assess NV strategies during summer in high-rise buildings [307], to assess the impact of window parameters on indoor NV [308] or to assess the impact of different strategies such us orientation, WWR and coating colour on NV [309]. The coupled simulations between IESVE software with MicroFlo extension (CFD simulator) was carried out by Bay et al., [310], to assess thermal and energy performance in different ventilation scenarios (i.e. no NV; fully open windows and doors; night ventilation) in a Church. A triple combination was developed by Abbas and Gursel Dino [311], to analyse the effectiveness of NV in preventing infection risk for COVID-19 through the indoor air of a NV room. With this goal, the authors execute Open Foam to perform CFD simulation, EnergyPlus to carry out energy modelling and simulations, and NIST CONTAM to compute contamination analyses. The Analysis System ANSYS Fluent and EnergyPlus programs were used by Liu et al. [312] to evaluate the effectiveness of transom window in reducing cooling energy use in high-rise residential buildings. In this study, CFD simulation allow to obtain an accurate prediction of achievable ventilation rates and indoor air velocities, while energy simulation predicted the hour-by-hour indoor air temperatures and potential cooling energy saving. The indoor thermal comfort was evaluated based on adaptive comfort models and using the predicted parameters (air velocities and temperatures). The combination of Fluent and EnergyPlus was led by Albuquerque et al. [313], to analyse night-time NV of a large atrium in no-residential buildings. The coupling program between Integrated Environmental Solutions Virtual Environment (IESVE) for natural ventilation and daylight simulations and Design-Builder for thermal comfort and energy consumption simulations was used by Zoure and Genovese [238]. The results obtained facilitated the investigation of NV and daylighting improvements in office buildings in order to maximize thermal comfort and energy efficiency. Carrilho da Graça et al, [314] used PHOENICS and Design Builder to analyse NV in shopping mall and identify thermal problems.

4.4.4. Simulations tools combined with multi-objective optimization methods (MOM)

It should be also noted that multi-objective optimization methods combined with simulation tools, such as CFD or BES, has been investigated to find optimal configurations of NV. Indeed, these tools have also been applied in the process of optimising the design of buildings with passive cooling measures. For example, the Non-dominant Sorting Genetic Algorithm II (NSGA-II) and EnergyPlus were developed by Baba et al. ([315]) to identify those school building characteristics that minimise overheating hours, heating energy and artificial lighting energy through a multi-objective optimisation process (the selected optimisation variables were thermal properties of building envelope components, exterior and interior shading options, WWR, Window Opening Ratio (WOR), night cooling and cool roof). This combination was also used in

previous studies to investigate the optimization of passive cooling in vernacular houses in order to minimize CO₂ emissions and cooling load (the optimization parameters were WWR, site orientation, infiltration level, local shading type, external wall construction type, roof type, ground floor construction, glazing type) [316] or to investigate how buildings characteristics (i.e. different glazing types, windows area, building orientation, insulation, etc.) influence NV in a singlefamily house [317]. Liu et al. [237] also proposed a multi-objective optimization approach based on NSGA-II algorithm to reduce lighting and AC energy consumption by using NV, cool roofs and solar shading in residential buildings in China. Energy simulations were conducted using DesignBuilder and the parameters tested according to energy consumption were: orientation, roof albedo, roof insulation thickness, exterior wall insulation thickness, WWR, window heat transfer coefficient, window solar heat gain coefficient, ratio of window height to balcony depth, thermal mass and infiltration air mass flow coefficient. Harkouss et al. [318] implemented a methodology for passive design optimization of low energy buildings and evaluated different ventilation strategies in different climates using the NSGA-II algorithm. The impact of various ventilation strategies on heating demand, thermal comfort and CO₂ concentration was analysed by Grygierek and Ferdyn-Grygierek [239] based on a multi-objective process using Contam and EnergyPlus software in conjunction with the Python programming language. The opening and closing of windows as a function of temperature and CO₂ concentration were considered for the optimization of the NV in Single-Family House. Song et al. [319] investigated window-opening behaviour and simulated the optimal window opening time in residential houses located in 13 typical climate cities in Xinjiang (China) using aPMV index and EnergyPlus. The genetic algorithm included in the Galapagos module of the Grasshopper platform was executed for the optimization process.

4.4.5. Simulation of future scenarios

The scientific literature has also focused on the analysis of future scenarios that pose a challenge to society as consequence of climate change. Previous studies assessed temperature projections and subsequently, simulated the energy consumption using BES. For example, Pérez -Andreu et al. [320] explored a range of energy improvements measures applied to typical Mediterranean residential buildings under various climate-change scenario. They used the used the Global Circulation Models (CNRM-CM5 and MPI-ESM-LR), under two emission scenarios (RCP4.5 and RCP8.5) to predict temperature projections and used TRNSYS tools to carry out energy simulations. Bienvenido-Huertas et al. [321] evaluated adaptive strategies in NV social dwellings. They analysed the current scenario (2015, 2016, and 2017) and a climate change scenario (2030, 2040, 2050, 2060, 2070, 2080, 2090, and 2100) in 10 Spanish cities and subsequently assessed the thermal comfort hours based on the ASHRAE 55-2017 and EN 16798-1:2019 standards. The EPW files for future scenario were generated with METEONORM and incorporated into EnergyPlus for the simulation process. Fiorito et al. [322] put their efforts on the adaptation of users in a historic building, assuming five weather scenarios: typical meteorological year, current extreme weather year (2018) and future meteorological conditions (period 2011–2040 (Average 2020), period2041–2070 (Average 2050), and period 2071-2100 (Average 2080)). Future meteorological conditions were derived using the statistical downscaling method in EnergyPlus. Thapa [323] investigated the risk of overheating in low-rise NV residential buildings in northeast India. Thapa used EnergyPlus software to simulate indoor thermal comfort in future climatic condition and the CCWorldWeatherGen tool to generate TMY files for the years 2050 and 2080 by morphing the present-TMY weather file. Pajek and Košir [324] took advantage from this same set of tools to simulate long-term energy performance in single-family residential buildings under representative European climates for the periods "2020" (2011-2040), "2050" (2041-2070) and "2080" (2071-2100). Similar investigations were carried out in conjunction with the same tools in MM office buildings in different locations, such as United States and Canada [325,326] or residential MM buildings in Argentina [327].

Although NV can contribute significantly to improving indoor thermal comfort conditions, previous studies showed that its effectiveness may be limited in future scenarios as a result of climate change. In fact, several authors claim that the increase in temperature in warm climatic regions will lead to longer hours of thermal discomfort in the summer months [321]. The analysis of future scenarios shows that the use of NV can improve the indoor thermal conditions during mid-seasons (spring and autumn), while this ventilation strategy should be complemented with other passive techniques or with the use of AC system during the summer months [327]. Therefore, the development of new techniques and the use of renewable energies to replace conventional energy is crucial to ensure the sustainability of the building stock in warm climate zones, while guaranteeing the thermal comfort of the building occupants.

4.4.6. Future perspectives

Finally, future research should focus on the following issues:

- Few studies use CFD to study IAQ during hot periods. Most of the reviewed studies employ simulation software for the analysis of energy or thermal comfort in different types of buildings or in the use of different passive cooling strategies. More studies are needed to understand in detail the ventilation strategies and IAQ in this scenario in hot/warm regions [196].
- Future research could examine the applicability of using BIM to address ventilation and energy performance in NV and MM buildings. BIM tools provide detailed information on the construction geometry and materials that can be exported to feed the generation of computational domains in CFD and BES models. Studies focused on interoperability and file sharing are needed.
- Generative design and BES/CFD systems. Technological evolution in the construction sector has led to the emergence of new tools such as the use of multi-objective optimization or generative design. Generative design and BES or CFD are tools that can be used to improve the ventilation and energy efficiency of buildings. Generative design is a computer-aided design process that uses algorithms to generate optimal design solutions. Future

research should study the use of these tools combined with BES and CFD tools to obtain better results, as well as improve the time consumption in the simulation process.

5. Discussion and Conclusions

The main contribution of this review article was the development of a detailed framework on the use of NV and MM in hot climate zones. The review carried out a bibliometric analysis from 1928 to 2023, to identify the main clusters that define the research area. Subsequently, publications of the last 20 years (2000-2023) were analysed, distinguishing among: (i) thermal comfort; (ii) energy efficiency; (iii) IAQ; and (iv) simulation. The main results of the review are reported below:

- The use of NV and MM is effective in achieving reductions in energy consumption, especially those based on night ventilation. Although this strategy significantly decreases the interior temperature, it can compromise the quality of sleep-in residential buildings due to the noise. In addition, NV and MM may have limited use due to outside contamination. Finally, the combination of NV and MM with other technical measures (i.e. thermal properties of windows) should be addressed to know in detail the most appropriate energy improvement strategies.
- No homogenization of the studies in building types. Currently, most of the studies have been carried out in residential and office buildings. It is necessary to expand knowledge in other typologies (i.e. nursing homes). Likewise, it is also required to evaluate more climatic zones in order to ensure the reproducibility of the methods and the respective results. In this sense, many studies were developed in Australia, China and India. However, there are few studies conducted in southern Europe and South America.
- The physiological adaptation should be evaluated in-depth. Previous studies have shown that the adaptation may not be influenced by short-term thermal history. However, the impact of user's behaviour on the potential effectiveness of NV and MM should be assessed.
- Effect of climate change and urban heat islands. Few studies have been detected related to the effects of climate change and heat islands in NV and MM. Hence, further research should be developed in this topic. In addition, improvement strategies (such as passive cooling to avoid overheating of buildings) should be tested.
- Ventilation strategies and IAQ need to be further explored. Despite a wide variety of studies (especially after the Covid-19 lockdown), most of them have an IAQ enhancement approach with basic ventilation strategies. Nevertheless, some aspects require an in-depth analysis in warm climate zones, such as: the implementation of special systems (i.e. air filters with solar chimneys and wind catchers); the effectiveness of MM in improving IAQ; the seasonal variability of the IAQ and the relationship with thermal comfort.
- For simulation and analysis purposes, it is detected that few studies use CFD to evaluate IAQ in warm months. Most of these researches focus on energy or thermal domains. Likewise, the use of BIM and generative design for NV and MM analyses should be promoted to optimize the results of the simulations.

To conclude, the review has shown the need of a holistic analysis about the potential of NV and MM in hot areas. Partial analysis are common in the state of the art, but they do not consider other dimensions of the topic that may affect the effectiveness of the strategies. Likewise, the studies are limited to certain regions, so the replicability of the methods can lead to variable results. Therefore, it is necessary to broaden the perspective of the analysis of NV and MM in warm areas. The use of these techniques can play a prominent role in the face of climate change and compliance with the Sustainable Development Goals. The reported information may be used by architects, engineers, researchers, and authorities related to the built environment in hot climate zones.

References

- [1] T. Letcher, Climate change: observed impacts on planet Earth, Elsevier, 2021.
- [2] L.H. Connor, Climate Change and Anthropos: Planet, people and places, Routledge, 2017.
- [3] J. Huang, G. Zhang, Y. Zhang, X. Guan, Y. Wei, R. Guo, Global desertification vulnerability to climate change and human activities, Land Degrad Dev. 31 (2020) 1380–1391.
- [4] European Commission, A Roadmap for moving to a competitive low carbon economy in 2050, Brussels, Belgium, 2011.
- [5] H. Huang, H. Wang, Y.J. Hu, C. Li, X. Wang, The development trends of existing building energy conservation and emission reduction—A comprehensive review, Energy Reports. 8 (2022) 13170–13188. https://doi.org/10.1016/j.egyr.2022.10.023.
- [6] J.D. Clark, B.D. Less, S.M. Dutton, I.S. Walker, M.H. Sherman, Efficacy of occupancy-based smart ventilation control strategies in energy-efficient homes in the United States, Build Environ. 156 (2019) 253–267. https://doi.org/10.1016/j.buildenv.2019.03.002.
- [7] Y. Li, P. Heiselberg, Analysis Methods for Natural and Hybrid Ventilation a Critical Literature Review and Recent Developments, International Journal of Ventilation. 1 (2003) 3–20. https://doi.org/10.1080/14733315.2003.11683640.
- [8] S. Omrani, V. Garcia-Hansen, B.R. Capra, R. Drogemuller, On the effect of provision of balconies on natural ventilation and thermal comfort in high-rise residential buildings, Build Environ. 123 (2017) 504–516. https://doi.org/10.1016/j.buildenv.2017.07.016.
- [9] K. Hiyama, L. Glicksman, Preliminary design method for naturally ventilated buildings using target air change rate and natural ventilation potential maps in the United States, Energy. 89 (2015) 655–666. https://doi.org/10.1016/j.energy.2015.06.026.
- [10] J.A. Castillo, G. Huelsz, A methodology to evaluate the indoor natural ventilation in hot climates: Heat Balance Index, Build Environ. 114 (2017) 366–373. https://doi.org/10.1016/j.buildenv.2016.12.027.

- [11] J. Kim, F. Tartarini, T. Parkinson, P. Cooper, R. de Dear, Thermal comfort in a mixed-mode building: Are occupants more adaptive?, Energy Build. 203 (2019). https://doi.org/10.1016/j.enbuild.2019.109436.
- [12] J. Kim, R. de Dear, Is mixed-mode ventilation a comfortable low-energy solution? A literature review, Build Environ. 205 (2021) 108215. https://doi.org/10.1016/j.buildenv.2021.108215.
- [13] R. Mateus, M.C. Pereira, A. Pinto, Energy & Buildings Natural ventilation of large air masses: Experimental and numerical techniques review, 291 (2023). https://doi.org/10.1016/j.enbuild.2023.113120.
- [14] H.Y. Zhong, Y. Sun, J. Shang, F.P. Qian, F.Y. Zhao, H. Kikumoto, C. Jimenez-Bescos, X. Liu, Single-sided natural ventilation in buildings: a critical literature review, Build Environ. 212 (2022) 108797. https://doi.org/10.1016/j.buildenv.2022.108797.
- [15] S. Ferrari, T. Blázquez, R. Cardelli, G. Puglisi, R. Suárez, L. Mazzarella, Ventilation strategies to reduce airborne transmission of viruses in classrooms: A systematic review of scientific literature, Build Environ. 222 (2022). https://doi.org/10.1016/j.buildenv.2022.109366.
- [16] N. Izadyar, W. Miller, Ventilation strategies and design impacts on indoor airborne transmission: A review, Build Environ. 218 (2022) 109158. https://doi.org/10.1016/j.buildenv.2022.109158.
- [17] T. Ahmed, P. Kumar, L. Mottet, Natural ventilation in warm climates: The challenges of thermal comfort, heatwave resilience and indoor air quality, Renewable and Sustainable Energy Reviews. 138 (2021) 110669. https://doi.org/10.1016/j.rser.2020.110669.
- [18] Z. Jiang, T. Kobayashi, T. Yamanaka, M. Sandberg, Energy & Buildings A literature review of cross ventilation in buildings, Energy Build. 291 (2023) 113143. https://doi.org/10.1016/j.enbuild.2023.113143.
- [19] M.E. Falagas, E.I. Pitsouni, G.A. Malietzis, G. Pappas, Comparison of PubMed, Scopus, Web of Science, and Google Scholar: strengths and weaknesses, The FASEB Journal. 22 (2008) 338–342. https://doi.org/10.1096/fj.07-9492lsf.
- [20] E. Borri, G. Zsembinszki, L.F. Cabeza, Recent developments of thermal energy storage applications in the built environment: A bibliometric analysis and systematic review, Appl Therm Eng. 189 (2021) 116666. https://doi.org/10.1016/j.applthermaleng.2021.116666.
- [21] B. Tejedor, E. Lucchi, D. Bienvenido-Huertas, I. Nardi, Non-destructive techniques (NDT) for the diagnosis of heritage buildings: Traditional procedures and futures perspectives, Energy Build. 263 (2022) 112029. https://doi.org/10.1016/j.enbuild.2022.112029.
- [22] R. Ren, W. Hu, J. Dong, B. Sun, Y. Chen, Z. Chen, A systematic literature review of green and sustainable logistics: Bibliometric analysis, research trend and knowledge taxonomy, Int J Environ Res Public Health. 17 (2020). https://doi.org/10.3390/ijerph17010261.
- [23] N. Andersen, V. Swami, Science mapping research on body image: A bibliometric review of publications in Body Image, 2004–2020, Body Image. 38 (2021) 106–119. https://doi.org/10.1016/j.bodyim.2021.03.015.
- [24] L.F. Cabeza, M. Chàfer, É. Mata, Comparative analysis of web of science and scopus on the energy efficiency and climate impact of buildings, Energies (Basel). 13 (2020). https://doi.org/10.3390/en13020409.
- [25] M. Angrisano, F. Fabbrocino, P. Iodice, L.F. Girard, The evaluation of historic building energy retrofit projects through the life cycle assessment, Applied Sciences (Switzerland). 11 (2021). https://doi.org/10.3390/app11157145.
- [26] I. Scoones, P. Forster, The International Response to Highly Pathogenic Avian Influenza: Science, Policy and Politics, STEPS Working Paper 10 STEPS Centre. (2008) 1–55. https://opendocs.ids.ac.uk/opendocs/bitstream/handle/123456789/2316/The International Response to Highly Pathogenic.pdf?sequence=1%0Ahttp://r4d.dfid.gov.uk/Output/178288/Default.aspx.
- [27] A. Wilkinson, M. Parker, F. Martineau, M. Leach, Engaging 'communities': Anthropological insights from the west African ebola epidemic, Philosophical Transactions of the Royal Society B: Biological Sciences. 372 (2017). https://doi.org/10.1098/rstb.2016.0305.
- [28] M. Leach, H. MacGregor, I. Scoones, A. Wilkinson, Post-pandemic transformations: How and why COVID-19 requires us to rethink development, World Dev. 138 (2021) 105233. https://doi.org/10.1016/j.worlddev.2020.105233.
- [29] ASHRAE Standard 55-2020, Thermal Environmental Conditions for Human Occupancy, (n.d.).
- [30] ISO 7730:2005., Ergonomics of the thermal environment Analytical determination and interpretation of thermal comfort using calculation of the PMV and PPD indices and local thermal comfort criteria, (n.d.).
- [31] G.S. Brager, R.J. De Dear, Thermal adaptation in the built environment: a literature review, Energy Build. 27 (1998) 83–96. https://doi.org/10.1016/S0378-7788(97)00053-4.
- [32] L. Yang, H. Yan, J.C. Lam, Thermal comfort and building energy consumption implications A review, Appl Energy. 115 (2014) 164–173. https://doi.org/10.1016/J.APENERGY.2013.10.062.
- [33] M. Deuble, R. De Dear, Mixed-Mode buildings: A double standard in comfort, in: 12th International Conference on Indoor Air Quality and Climate 2011, Macquarie University, North Ryde NSW 2109, Australia, 2011: pp. 1313–1318. https://www.scopus.com/inward/record.uri?eid=2-s2.0-84880544428&partnerID=40&md5=32102da2310ea8ce8e891ad0daba61d6.

 [34] M.P. Deuble, R.J. de Dear, Mixed-mode buildings: A double standard in occupants' comfort expectations, Build Environ. 54 (2012) 53–60. https://doi.org/10.1016/j.buildenv.2012.01.021.

- [35] N. Bhandari, S. Tadepalli, P. Gopalakrishnan, Influence of non-uniform distribution of fan-induced air on thermal comfort conditions in university classrooms in warm and humid climate, India, Build Environ. 238 (2023). https://doi.org/10.1016/j.buildenv.2023.110373.
- [36] A. Sharma, A. Kumar, Adaptive thermal comfort of residential buildings in the composite climatic region of India: a field study, Architectural Engineering and Design Management. (2023). https://doi.org/10.1080/17452007.2023.2201416.

- [37] C. Liu, H. Xie, H.M. Ali, J. Liu, Evaluation of Passive Cooling and Thermal Comfort in Historical Residential Buildings in Zanzibar, Buildings. 12 (2022). https://doi.org/10.3390/buildings12122149.
- [38] N.K. Khambadkone, P. Madhumati, M.N. Ranganath, Thermal comfort evaluation in architectural studio classrooms – A summer study in a warm to moderate Indian climate, Indoor and Built Environment. 31 (2022) 2331–2365. https://doi.org/10.1177/1420326X221098552.
- [39] R. Rawal, Y. Shukla, V. Vardhan, S. Asrani, M. Schweiker, R. de Dear, V. Garg, J. Mathur, S. Prakash, S. Diddi, S. V Ranjan, A.N. Siddiqui, G. Somani, Adaptive thermal comfort model based on field studies in five climate zones across India, Build Environ. 219 (2022). https://doi.org/10.1016/j.buildenv.2022.109187.
- [40] M.T. Baquero Larriva, A.S. Mendes, N. Forcada, The effect of climatic conditions on occupants' thermal comfort in naturally ventilated nursing homes, Build Environ. 214 (2022). https://doi.org/10.1016/j.buildenv.2022.108930.
- [41] C. Koranteng, B. Simons, K.G. Abrokwa, Thermal State, Potential of Natural and Fan-Induced Ventilation in Enhancing Comfort of Nine Office Buildings in Southern Ghana, Journal of Architectural Engineering. 28 (2022). https://doi.org/10.1061/(ASCE)AE.1943-5568.0000521.
- [42] C. Heracleous, A. Michael, A. Savvides, C. Hayles, Climate change resilience of school premises in Cyprus: An examination of retrofit approaches and their implications on thermal and energy performance, Journal of Building Engineering. 44 (2021). https://doi.org/10.1016/j.jobe.2021.103358.
- [43] N. Forcada, M. Gangolells, M. Casals, B. Tejedor, M. Macarulla, K. Gaspar, Field study on adaptive thermal comfort models for nursing homes in the Mediterranean climate, Energy Build. 252 (2021). https://doi.org/10.1016/j.enbuild.2021.111475.
- [44] A. Bassoud, H. Khelafi, A.M. Mokhtari, A. Bada, Evaluation of summer thermal comfort in arid desert areas. Case study: Old adobe building in Adrar (South of Algeria), Build Environ. 205 (2021). https://doi.org/10.1016/j.buildenv.2021.108140.
- [45] X. Wang, L. Yang, S. Gao, S. Zhao, Y. Zhai, Thermal comfort in naturally ventilated university classrooms: A seasonal field study in Xi'an, China, Energy Build. 247 (2021). https://doi.org/10.1016/j.enbuild.2021.111126.
- [46] H. Ikeda, T. Nakaya, A. Nakagawa, Y. Maeda, An investigation of indoor thermal environment in semi-cold region in Japan – Validity of thermal predictive indices in Nagano during the summer season, Journal of Building Engineering. 35 (2021). https://doi.org/10.1016/j.jobe.2020.101897.
- [47] G. Guevara, G. Soriano, I. Mino-Rodriguez, Thermal comfort in university classrooms: An experimental study in the tropics, Build Environ. 187 (2021). https://doi.org/10.1016/j.buildenv.2020.107430.
- [48] C. Munonye, Y. Ji, Evaluating the perception of thermal environment in naturally ventilated schools in a warm and humid climate in Nigeria, Building Services Engineering Research and Technology. 42 (2021) 5–25. https://doi.org/10.1177/0143624420911148.
- [49] M.S.J. Talukdar, T.H. Talukdar, M.K. Singh, M.A. Baten, M.S. Hossen, Status of thermal comfort in naturally ventilated university classrooms of Bangladesh in hot and humid summer season, Journal of Building Engineering. 32 (2020). https://doi.org/10.1016/j.jobe.2020.101700.
- [50] M. Jowkar, R. de Dear, J. Brusey, Influence of long-term thermal history on thermal comfort and preference, Energy Build. 210 (2020). https://doi.org/10.1016/j.enbuild.2019.109685.
- [51] S. Sharifi, W. Saman, A. Alemu, Identification of overheating in the top floors of energy-efficient multilevel dwellings, Energy Build. 204 (2019). https://doi.org/10.1016/j.enbuild.2019.109452.
- [52] Y. Zhai, A. Honnekeri, M. Pigman, M. Fountain, H. Zhang, X. Zhou, E. Arens, Use of adaptive control and its effects on human comfort in a naturally ventilated office in Alameda, California, Energy Build. 203 (2019). https://doi.org/10.1016/j.enbuild.2019.109435.
- [53] A. Jindal, Investigation and analysis of thermal comfort in naturally ventilated secondary school classrooms in the composite climate of India, Archit Sci Rev. 62 (2019) 466–484. https://doi.org/10.1080/00038628.2019.1653818.
- [54] Z. Wu, N. Li, P. Wargocki, J. Peng, J. Li, H. Cui, Field study on thermal comfort and energy saving potential in 11 split air-conditioned office buildings in Changsha, China, Energy. 182 (2019) 471–482. https://doi.org/10.1016/j.energy.2019.05.204.
- [55] M. Khoshbakht, Z. Gou, F. Zhang, A pilot study of thermal comfort in subtropical mixed-mode higher education office buildings with different change-over control strategies, Energy Build. 196 (2019) 194–205. https://doi.org/10.1016/j.enbuild.2019.05.030.
- [56] Z. Wu, N. Li, P. Wargocki, J. Peng, J. Li, H. Cui, Adaptive thermal comfort in naturally ventilated dormitory buildings in Changsha, China, Energy Build. 186 (2019) 56–70. https://doi.org/10.1016/j.enbuild.2019.01.029.
- [57] L.A. López-Pérez, J.J. Flores-Prieto, C. Ríos-Rojas, Adaptive thermal comfort model for educational buildings in a hothumid climate, Build Environ. 150 (2019) 181–194. https://doi.org/10.1016/j.buildenv.2018.12.011.
- [58] S.S.Y. Lau, J. Zhang, Y. Tao, A comparative study of thermal comfort in learning spaces using three different ventilation strategies on a tropical university campus, Build Environ. 148 (2019) 579–599. https://doi.org/10.1016/j.buildenv.2018.11.032.
- [59] S. Kumar, M.K. Singh, A. Mathur, J. Mathur, S. Mathur, Evaluation of comfort preferences and insights into behavioural adaptation of students in naturally ventilated classrooms in a tropical country, India, Build Environ. 143 (2018) 532– 547. https://doi.org/10.1016/j.buildenv.2018.07.035.
- [60] C. Xu, S. Li, X. Zhang, S. Shao, Thermal comfort and thermal adaptive behaviours in traditional dwellings: A case study in Nanjing, China, Build Environ. 142 (2018) 153–170. https://doi.org/10.1016/j.buildenv.2018.06.006.
- [61] B. Hamzah, Z. Gou, R. Mulyadi, S. Amin, Thermal comfort analyses of secondary school students in the tropics, Buildings. 8 (2018). https://doi.org/10.3390/buildings8040056.

- [62] M.K. Singh, S. Kumar, R. Ooka, H.B. Rijal, G. Gupta, A. Kumar, Status of thermal comfort in naturally ventilated classrooms during the summer season in the composite climate of India, Build Environ. 128 (2018) 287–304. https://doi.org/10.1016/j.buildenv.2017.11.031.
- [63] H. Liu, Y. Wu, B. Li, Y. Cheng, R. Yao, Seasonal variation of thermal sensations in residential buildings in the Hot Summer and Cold Winter zone of China, Energy Build. 140 (2017) 9–18. https://doi.org/10.1016/j.enbuild.2017.01.066.
- [64] S.A. Damiati, S.A. Zaki, H.B. Rijal, S. Wonorahardjo, Field study on adaptive thermal comfort in office buildings in Malaysia, Indonesia, Singapore, and Japan during hot and humid season, Build Environ. 109 (2016) 208–223. https://doi.org/10.1016/j.buildenv.2016.09.024.
- [65] M.S. Mustapa, S.A. Zaki, H.B. Rijal, A. Hagishima, M.S.M. Ali, Thermal comfort and occupant adaptive behaviour in Japanese university buildings with free running and cooling mode offices during summer, Build Environ. 105 (2016) 332–342. https://doi.org/10.1016/j.buildenv.2016.06.014.
- [66] S. Manu, Y. Shukla, R. Rawal, L.E. Thomas, R. de Dear, Field studies of thermal comfort across multiple climate zones for the subcontinent: India Model for Adaptive Comfort (IMAC), Build Environ. 98 (2016) 55–70. https://doi.org/10.1016/j.buildenv.2015.12.019.
- [67] S. Dhaka, J. Mathur, G. Brager, A. Honnekeri, Assessment of thermal environmental conditions and quantification of thermal adaptation in naturally ventilated buildings in composite climate of India, Build Environ. 86 (2015) 17–28. https://doi.org/10.1016/j.buildenv.2014.11.024.
- [68] H.B. Rijal, Investigation of comfort temperature and occupant behavior in Japanese houses during the hot and humid season, Buildings. 4 (2014) 437–452. https://doi.org/10.3390/buildings4030437.
- [69] M. Indraganti, R. Ooka, H.B. Rijal, G.S. Brager, Adaptive model of thermal comfort for offices in hot and humid climates of India, Build Environ. 74 (2014) 39–53. https://doi.org/10.1016/j.buildenv.2014.01.002.
- [70] M. Indraganti, R. Ooka, H.B. Rijal, Field investigation of comfort temperature in Indian office buildings: A case of Chennai and Hyderabad, Build Environ. 65 (2013) 195–214. https://doi.org/10.1016/j.buildenv.2013.04.007.
- [71] G. Bravo, E. González, Thermal comfort in naturally ventilated spaces and under indirect evaporative passive cooling conditions in hot-humid climate, Energy Build. 63 (2013) 79–86. https://doi.org/10.1016/j.enbuild.2013.03.007.
- [72] E. Rajasekar, A. Ramachandraiah, A study on thermal parameters in residential buildings associated with hot humid environments, Archit Sci Rev. 54 (2011) 23–38. https://doi.org/10.3763/asre.2009.0082.
- [73] M. Indraganti, Adaptive use of natural ventilation for thermal comfort in Indian apartments, Build Environ. 45 (2010) 1490–1507. https://doi.org/10.1016/j.buildenv.2009.12.013.
- [74] M. Indraganti, K.D. Rao, Effect of age, gender, economic group and tenure on thermal comfort: A field study in residential buildings in hot and dry climate with seasonal variations, Energy Build. 42 (2010) 273–281. https://doi.org/10.1016/j.enbuild.2009.09.003.
- [75] M. Indraganti, Using the adaptive model of thermal comfort for obtaining indoor neutral temperature: Findings from a field study in Hyderabad, India, Build Environ. 45 (2010) 519–536. https://doi.org/10.1016/j.buildenv.2009.07.006.
- [76] M. Indraganti, Thermal comfort in naturally ventilated apartments in summer: Findings from a field study in Hyderabad, India, Appl Energy. 87 (2010) 866–883. https://doi.org/10.1016/j.apenergy.2009.08.042.
- [77] W. Yang, G. Zhang, Thermal comfort in naturally ventilated and air-conditioned buildings in humid subtropical climate zone in China, Int J Biometeorol. 52 (2008) 385–398. https://doi.org/10.1007/s00484-007-0133-4.
- [78] N.H. Wong, S.S. Khoo, Thermal comfort in classrooms in the tropics, Energy Build. 35 (2003) 337–351. https://doi.org/10.1016/S0378-7788(02)00109-3.
- [79] C. Zepeda-Gil, S. Natarajan, Thermal comfort in naturally ventilated dwellings in the central Mexican plateau, Build Environ. 211 (2022). https://doi.org/10.1016/j.buildenv.2021.108713.
- [80] G. Ramos, R. Lamberts, K.C.F.J. Abrahão, F.B. Bandeira, C.F. Barbosa Teixeira, M. Brito de Lima, E.E. Broday, A.P.A.S. Castro, L. de Queiroz Leal, R. De Vecchi, L.D.M. De Zorzi, K.A. Dornelles, S. Duarte, R.G. Faisca, M. Fontenelle, T. Freitas, C. Gregorio Atem, G. Grigoletti, L. Maciel, C.M. de Moraes, C.M. Mueller, T. Sartori, B. Silva, C. Silva, M. Vilar da Silva, A.L. Silveira, R.V.G. Souza, A.A. Xavier, Adaptive behaviour and air conditioning use in Brazilian residential buildings, Building Research and Information. 49 (2021) 496–511. https://doi.org/10.1080/09613218.2020.1804314.
- [81] J. Morey, A. Beizaee, A. Wright, An investigation into overheating in social housing dwellings in central England, Build Environ. 176 (2020). https://doi.org/10.1016/j.buildenv.2020.106814.
- [82] C. Buonocore, R. De Vecchi, V. Scalco, R. Lamberts, Thermal preference and comfort assessment in air-conditioned and naturally-ventilated university classrooms under hot and humid conditions in Brazil, Energy Build. 211 (2020). https://doi.org/10.1016/j.enbuild.2020.109783.
- [83] A. Dhalluin, K. Limam, Comparison of natural and hybrid ventilation strategies used in classrooms in terms of indoor environmental quality, comfort and energy savings, Indoor and Built Environment. 23 (2014) 527–542. https://doi.org/10.1177/1420326X12464077.
- [84] C. Koranteng, B. Simons, C. Essel, Climate responsive buildings: a comfort assessment of buildings on KNUST campus, Kumasi, Journal of Engineering, Design and Technology. 17 (2019) 862–877. https://doi.org/10.1108/JEDT-03-2019-0054.
- [85] M.K. Nematchoua, P. Ricciardi, J.A. Orosa, S. Asadi, R. Choudhary, Influence of indoor environmental quality on the self-estimated performance of office workers in the tropical wet and hot climate of Cameroon, Journal of Building Engineering. 21 (2019) 141–148. https://doi.org/10.1016/j.jobe.2018.10.007.

- [86] E. Kyritsi, A. Michael, An assessment of the impact of natural ventilation strategies and window opening patterns in office buildings in the mediterranean basin, Build Environ. 175 (2020). https://doi.org/10.1016/j.buildenv.2019.106384.
- [87] Y. HE, F. HAN, L. ZHAO, S. LI, Investigation of the indoor environment of a passive house office building under cold climate in China, Indoor and Built Environment. 32 (2023) 603–623. https://doi.org/10.1177/1420326X221126011.
- [88] D. Lai, Y. Qi, J. Liu, X. Dai, L. Zhao, S. Wei, Ventilation behavior in residential buildings with mechanical ventilation systems across different climate zones in China, Build Environ. 143 (2018) 679–690. https://doi.org/10.1016/j.buildenv.2018.08.006.
- [89] D. Lai, S. Jia, Y. Qi, J. Liu, Window-opening behavior in Chinese residential buildings across different climate zones, Build Environ. 142 (2018) 234–243. https://doi.org/10.1016/j.buildenv.2018.06.030.
- [90] R. KC, H.B. Rijal, M. Shukuya, K. Yoshida, An in-situ study on occupants' behaviors for adaptive thermal comfort in a Japanese HEMS condominium, Journal of Building Engineering. 19 (2018) 402–411. https://doi.org/10.1016/j.jobe.2018.05.013.
- [91] L. Daniel, 'We like to live in the weather': Cooling practices in naturally ventilated dwellings in Darwin, Australia, Energy Build. 158 (2018) 549–557. https://doi.org/10.1016/j.enbuild.2017.10.031.
- [92] H.-H. Liang, T.-P. Lin, R.-L. Hwang, Linking occupants' thermal perception and building thermal performance in naturally ventilated school buildings, Appl Energy. 94 (2012) 355–363. https://doi.org/10.1016/j.apenergy.2012.02.004.
- [93] P. Aparicio-Ruiz, E. Barbadilla-Martín, J. Guadix, J. Muñuzuri, A field study on adaptive thermal comfort in Spanish primary classrooms during summer season, Build Environ. 203 (2021). https://doi.org/10.1016/j.buildenv.2021.108089.
- [94] C. Sun, R. Zhang, S. Sharples, Y. Han, H. Zhang, A longitudinal study of summertime occupant behaviour and thermal comfort in office buildings in northern China, Build Environ. 143 (2018) 404–420. https://doi.org/10.1016/j.buildenv.2018.07.004.
- [95] J. Hou, Y. Sun, Y. Song, J. Kim, T. Parkinson, R. de Dear, Study on adaptive comfort behaviours in mixed-mode residential buildings in Tianjin, China, Indoor and Built Environment. 31 (2022) 777–787. https://doi.org/10.1177/1420326X211032170.
- [96] K. Zhong, H. Fu, Y. Kang, X. Peng, Indoor thermal conditions and the potential of energy conservation of naturally ventilated rooms in summer, China, Energy Build. 55 (2012) 183–188. https://doi.org/10.1016/j.enbuild.2012.08.027.
- [97] C.-P. Chen, R.-L. Hwang, W.-M. Shih, Effect of fee-for-service air-conditioning management in balancing thermal comfort and energy usage, Int J Biometeorol. 58 (2014) 1941–1950. https://doi.org/10.1007/s00484-014-0796-6.
- [98] H. Yan, Z. Sun, F. Shi, G. Yuan, M. Dong, M. Wang, Thermal response and thermal comfort evaluation of the split air conditioned residential buildings, Build Environ. 221 (2022). https://doi.org/10.1016/j.buildenv.2022.109326.
- [99] R. De Dear, J. Kim, C. Candido, M. Deuble, Adaptive thermal comfort in australian school classrooms, Building Research and Information. 43 (2015) 383–398. https://doi.org/10.1080/09613218.2015.991627.
- [100] M.P. Deuble, R.J. de Dear, Is it hot in here or is it just me? Validating the post-occupancy evaluation, Intelligent Buildings International. 6 (2014) 112–134. https://doi.org/10.1080/17508975.2014.883299.
- [101] R.F. Rupp, J. Kim, R. de Dear, E. Ghisi, Associations of occupant demographics, thermal history and obesity variables with their thermal comfort in air-conditioned and mixed-mode ventilation office buildings, Build Environ. 135 (2018) 1–9. https://doi.org/10.1016/j.buildenv.2018.02.049.
- [102] R. De Vecchi, C. Candido, R. de Dear, R. Lamberts, Thermal comfort in office buildings: Findings from a field study in mixed-mode and fully-air conditioning environments under humid subtropical conditions, Build Environ. 123 (2017) 672–683. https://doi.org/10.1016/j.buildenv.2017.07.029.
- [103] A.A. Bagasi, J.K. Calautit, Experimental field study of the integration of passive and evaporative cooling techniques with Mashrabiya in hot climates, Energy Build. 225 (2020). https://doi.org/10.1016/j.enbuild.2020.110325.
- [104] T. Kubota, D.T.H. Chyee, Potential of passive cooling techniques for modern houses in the tropical climate of malaysia
 Analysis of the indoor thermal environment for various ventilation strategies, International Journal of Ventilation.
 9 (2010) 11–24. https://doi.org/10.1080/14733315.2010.11683863.
- [105] C. Heracleous, A. Michael, Assessment of overheating risk and the impact of natural ventilation in educational buildings of Southern Europe under current and future climatic conditions, Energy. 165 (2018) 1228–1239. https://doi.org/10.1016/j.energy.2018.10.051.
- [106] A. Michael, D. Demosthenous, M. Philokyprou, Natural ventilation for cooling in mediterranean climate: A case study in vernacular architecture of Cyprus, Energy Build. 144 (2017) 333–345. https://doi.org/10.1016/j.enbuild.2017.03.040.
- [107] D. Demosthenous, A. Michael, M. Philokyprou, Natural ventilation strategies for cooling purposes in the rural vernacular architecture of cyprus, Renewable Energy and Power Quality Journal. 1 (2016) 526–530. https://doi.org/10.24084/repqj14.381.
- [108] D.H.C. Toe, T. Kubota, Comparative assessment of vernacular passive cooling techniques for improving indoor thermal comfort of modern terraced houses in hot-humid climate of Malaysia, Solar Energy. 114 (2015) 229–258. https://doi.org/10.1016/j.solener.2015.01.035.
- [109] L. Moosavi, N. Mahyuddin, N. Ghafar, Atrium cooling performance in a low energy office building in the Tropics, a field study, Build Environ. 94 (2015) 384–394. https://doi.org/10.1016/j.buildenv.2015.06.020.

- [110] A.A. Elmualim, Verification of design calculations of a wind catcher/tower natural ventilation system with performance testing in a real building, International Journal of Ventilation. 4 (2006) 393–404. https://doi.org/10.1080/14733315.2005.11683717.
- [111] T. Psomas, P. Heiselberg, T. Lyme, K. Duer, Automated roof window control system to address overheating on renovated houses: Summertime assessment and intercomparison, Energy Build. 138 (2017) 35–46. https://doi.org/10.1016/j.enbuild.2016.12.019.
- [112] J. Ortiz, A. Fonseca, J. Salom, N. Garrido, P. Fonseca, V. Russo, Comfort and economic criteria for selecting passive measures for the energy refurbishment of residential buildings in Catalonia, Energy Build. 110 (2016) 195–210. https://doi.org/10.1016/j.enbuild.2015.10.022.
- [113] D.R. Rangaswamy, K. Ramamurthy, Evaluation of Eight Thermal Comfort Indices Based on Perception Survey for a Hot-Humid Climate through a Naturally Ventilated Apartment, Journal of Architectural Engineering. 27 (2021). https://doi.org/10.1061/(ASCE)AE.1943-5568.0000508.
- [114] N.M. Abd Rahman, L.C. Haw, A. Fazlizan, A. Hussin, M.S. Imran, Thermal comfort assessment of naturally ventilated public hospital wards in the tropics, Build Environ. 207 (2022). https://doi.org/10.1016/j.buildenv.2021.108480.
- [115] L. V de Abreu-Harbich, V.L.A. Chaves, M.C.G.O. Brandstetter, Evaluation of strategies that improve the thermal comfort and energy saving of a classroom of an institutional building in a tropical climate, Build Environ. 135 (2018) 257–268. https://doi.org/10.1016/j.buildenv.2018.03.017.
- [116] R. Daghigh, N.M. Adam, B.B. Sahari, Ventilation parameters and thermal comfort of naturally and mechanically ventilated offices, Indoor and Built Environment. 18 (2009) 113–122. https://doi.org/10.1177/1420326X09103013.
- [117] A. Wagner, E. Gossauer, C. Moosmann, Th. Gropp, R. Leonhart, Thermal comfort and workplace occupant satisfaction-Results of field studies in German low energy office buildings, Energy Build. 39 (2007) 758–769. https://doi.org/10.1016/j.enbuild.2007.02.013.
- [118] D. Sánchez-García, C. Rubio-Bellido, J.J.M. del Río, A. Pérez-Fargallo, Towards the quantification of energy demand and consumption through the adaptive comfort approach in mixed mode office buildings considering climate change, Energy Build. 187 (2019) 173–185. https://doi.org/10.1016/j.enbuild.2019.02.002.
- [119] K. Healey, Measurement and interpretation of thermal comfort in a highly adaptive mixed-mode building, Archit Sci Rev. 57 (2014) 207–214. https://doi.org/10.1080/00038628.2013.868782.
- [120] EUROSTAT statistic, Population structure and ageing. Available online: https://ec.europa.eu/eurostat/statistics-explained/index.php?title=Population_structure_and_ageing/es&oldid=364916]. (Accessed 16 June 2023), (n.d.).
- [121] D. Mongkolsawat, A. Marmot, M. Ucci, A comparison of perceived learning performance of Thai university students in fan-assisted naturally ventilated and air-conditioned classrooms, Intelligent Buildings International. 6 (2014) 93– 111. https://doi.org/10.1080/17508975.2014.893863.
- [122] F. Bougiatioti, E. Alexandrou, M. Katsaros, Sustainable refurbishment of existing, typical single-family residential buildings in Greece, International Journal of Building Pathology and Adaptation. (2023). https://doi.org/10.1108/IJBPA-09-2022-0157.
- [123] A.A. Jamaludin, H. Hussein, A.R. Mohd Ariffin, N. Keumala, A study on different natural ventilation approaches at a residential college building with the internal courtyard arrangement, Energy Build. 72 (2014) 340–352. https://doi.org/10.1016/j.enbuild.2013.12.050.
- [124] M. Aram, O. Abessi, Optimal design of green buildings using computational fluid dynamics and climate simulation tools, International Journal of Environmental Science and Technology. 17 (2020) 917–932. https://doi.org/10.1007/s13762-019-02403-6.
- [125] D. Bienvenido-Huertas, D. Sánchez-García, C. Rubio-Bellido, Analysing natural ventilation to reduce the cooling energy consumption and the fuel poverty of social dwellings in coastal zones, Appl Energy. 279 (2020). https://doi.org/10.1016/j.apenergy.2020.115845.
- [126] D. Bienvenido-Huertas, D. Sánchez-García, C. Rubio-Bellido, Influence of the RCP scenarios on the effectiveness of adaptive strategies in buildings around the world, Build Environ. 208 (2022). https://doi.org/10.1016/j.buildenv.2021.108631.
- [127] D. Bienvenido-Huertas, C. Rubio-Bellido, A. Pérez-Fargallo, J.A. Pulido-Arcas, Energy saving potential in current and future world built environments based on the adaptive comfort approach, J Clean Prod. 249 (2020). https://doi.org/10.1016/j.jclepro.2019.119306.
- [128] D. Bienvenido-Huertas, D. Sánchez-García, C. Rubio-Bellido, Influence of the RCP scenarios on the effectiveness of adaptive strategies in buildings around the world, Build Environ. 208 (2022). https://doi.org/10.1016/j.buildenv.2021.108631.
- [129] D. Bienvenido-Huertas, C. Rubio-Bellido, A. Pérez-Fargallo, J.A. Pulido-Arcas, Energy saving potential in current and future world built environments based on the adaptive comfort approach, J Clean Prod. 249 (2020). https://doi.org/10.1016/j.jclepro.2019.119306.
- [130] B.N. Kaudeer, S. Venkannah, Influence of the climate change on natural ventilation use in residential buildings in the tropical island of Mauritius, Indoor and Built Environment. (2023). https://doi.org/10.1177/1420326X231151857.
- [131] A. Rysanek, R. Nuttall, J. McCarty, Forecasting the impact of climate change on thermal comfort using a weighted ensemble of supervised learning models, Build Environ. 190 (2021). https://doi.org/10.1016/j.buildenv.2020.107522.
- [132] L. Pajek, M. Jevrić, I. Ćipranić, M. Košir, A multi-aspect approach to energy retrofitting under global warming: A case of a multi-apartment building in Montenegro, Journal of Building Engineering. 63 (2023). https://doi.org/10.1016/j.jobe.2022.105462.

- [133] X. Li, K. Huang, G. Feng, D. Jing, D. Liu, J. Li, Analysis on night ventilation effect of buildings with different energy consumption levels in Shenyang, in: L. A., O. T., K. R. (Eds.), 16th ROOMVENT Conference, ROOMVENT 2022, EDP Sciences, School of Municipal and Environmental Engineering, Shenyang Jianzhu University, Shenyang, 110168, China, 2022. https://doi.org/10.1051/e3sconf/202235601066.
- [134] K. Sun, W. Zhang, Z. Zeng, R. Levinson, M. Wei, T. Hong, Passive cooling designs to improve heat resilience of homes in underserved and vulnerable communities, Energy Build. 252 (2021). https://doi.org/10.1016/j.enbuild.2021.111383.
- [135] W. Feng, Y. Dongpeng, Study on Natural Ventilation Design Method of Residential Buildings in Summer and Winter Areas based on the computer technology, in: 2020 International Conference on Mechanical Automation and Computer Engineering, MACE 2020, IOP Publishing Ltd, Dalian University of Technology, Liaoning, 116035, China, 2021. https://doi.org/10.1088/1742-6596/1744/3/032242.
- [136] A.N. Zoure, P.V. Genovese, Implementing natural ventilation and daylighting strategies for thermal comfort and energy efficiency in office buildings in Burkina Faso, Energy Reports. 9 (2023) 3319–3342. https://doi.org/10.1016/j.egyr.2023.02.017.
- [137] A.N. Zoure, P. V Genovese, Development of Bioclimatic Passive Designs for Office Building in Burkina Faso, Sustainability (Switzerland). 14 (2022). https://doi.org/10.3390/su14074332.
- [138] G. Elshafei, S. Vilcekova, M. Zelenakova, A.M. Negm, Towards an adaptation of efficient passive design for thermal comfort buildings, Sustainability (Switzerland). 13 (2021) 1–23. https://doi.org/10.3390/su13179570.
- [139] H. Khalid, M.J. Thaheem, M.S.A. Malik, M.A. Musarat, W.S. Alaloul, Reducing cooling load and lifecycle cost for residential buildings: a case of Lahore, Pakistan, International Journal of Life Cycle Assessment. 26 (2021) 2355– 2374. https://doi.org/10.1007/s11367-021-02000-1.
- [140] R. Bo, Y. Shao, Y. Xu, Y. Yu, H. Guo, W.-S. Chang, Research on the Relationship between Thermal Insulation Thickness and Summer Overheating Risk: A Case Study in Severe Cold and Cold Regions of China, Buildings. 12 (2022). https://doi.org/10.3390/buildings12071032.
- [141] Z.A. Elhassan, Energy consumption performance using natural ventilation in dwelling design and CFD simulation in a hot dry climate: A case study in Sudan, Front Built Environ. 9 (2023). https://doi.org/10.3389/fbuil.2023.1145747.
- [142] S. Gunasagaran, E.S. Saw, T. Mari, S. Srirangam, V. Ng, Courtyard configuration to optimize shading, daylight and ventilation in a tropical terrace house using simulation, International Journal of Architectural Research: Archnet-IJAR. 17 (2023) 109–123. https://doi.org/10.1108/ARCH-12-2021-0354.
- [143] Q. Sun, Z. Luo, L. Bai, The Impact of Internal Courtyard Configuration on Thermal Performance of Long Strip Houses, Buildings. 13 (2023). https://doi.org/10.3390/buildings13020371.
- [144] B. Yan, X. Han, A. Malkawi, T.H. Dokka, P. Howard, J. Knowles, T. Hegli, K. Edwards, Comprehensive assessment of operational performance of coupled natural ventilation and thermally active building system via an extensive sensor network, Energy Build. 260 (2022). https://doi.org/10.1016/j.enbuild.2022.111921.
- [145] G. Elshafei, A. Negm, M. Bady, M. Suzuki, M.G. Ibrahim, Numerical and experimental investigations of the impacts of window parameters on indoor natural ventilation in a residential building, Energy Build. 141 (2017) 321–332. https://doi.org/10.1016/j.enbuild.2017.02.055.
- [146] M.W. Anwar, Z. Ali, A. Javed, E.U. Din, M. Sajid, Analysis of the effect of passive measures on the energy consumption and zero-energy prospects of residential buildings in Pakistan, Build Simul. 14 (2021) 1325–1342. https://doi.org/10.1007/s12273-020-0729-8.
- [147] F.-Z. El-Bichri, I. Sobhy, I. Bouchefra, B. Benhamou, H. Chehouani, M.O. Mghazli, Assessment of the impact of construction materials on the building's thermal behaviour and indoor thermal comfort in a hot and semi-arid climate, Advances in Building Energy Research. 16 (2022) 711–735. https://doi.org/10.1080/17512549.2022.2096692.
- [148] A. Albatayneh, Sensitivity analysis optimisation of building envelope parameters in a sub-humid Mediterranean climate zone, Energy Exploration and Exploitation. 39 (2021) 2080–2102. https://doi.org/10.1177/01445987211020432.
- [149] B. Akhozheya, S. Dagher, H. Slimani, Analyzing the Impact of High Thermal Mass and Natural Ventilation Strategies on Indoor Temperatures in an Extreme Climate, in: 2022 Advances in Science and Engineering Technology International Conferences, ASET 2022, Institute of Electrical and Electronics Engineers Inc., Polytechnic University of Milan, Department of Building Architectural Engineering, Milan, Italy, 2022. https://doi.org/10.1109/ASET53988.2022.9734804.
- [150] P. Bevilacqua, R. Bruno, J. Szyszka, D. Cirone, A. Rollo, Summer and winter performance of an innovative concept of Trombe wall for residential buildings, Energy. 258 (2022). https://doi.org/10.1016/j.energy.2022.124798.
- [151] N. Nasrollahi, P. Ghobadi, Field measurement and numerical investigation of natural cross-ventilation in high-rise buildings; Thermal comfort analysis, Appl Therm Eng. 211 (2022). https://doi.org/10.1016/j.applthermaleng.2022.118500.
- [152] T. Liu, X. Wang, W.L. Lee, Evaluating the effectiveness of transom window in reducing cooling energy use in high-rise residential buildings in Hong Kong, Journal of Building Engineering. 35 (2021). https://doi.org/10.1016/j.jobe.2020.102007.
- [153] V.M. Surendran, C. Irulappan, V. Jeyasingh, V. Ramalingam, Thermal Performance Assessment of Envelope Retrofits for Existing School Buildings in a Hot–Humid Climate: A Case Study in Chennai, India, Buildings. 13 (2023). https://doi.org/10.3390/buildings13041103.

- [154] M.J. Cook, Y. Shukla, R. Rawal, C. Angelopoulos, L. Caruggi-De-faria, D. Loveday, E. Spentzou, J. Patel, Integrating low energy cooling and ventilation strategies in Indian residences, Buildings and Cities. 3 (2022) 279–296. https://doi.org/10.5334/bc.197.
- [155] J. Ran, K. Xiong, M. Dou, H. Zhong, Y. Feng, M. Tang, Z. Yang, Effect of Window Openable Area and Shading on Indoor Thermal Comfort and Energy Efficiency in Residential Buildings with Various Operating Modes, Atmosphere (Basel). 13 (2022). https://doi.org/10.3390/atmos13122020.
- [156] D. Sánchez-García, C. Rubio-Bellido, J.J.M. del Río, A. Pérez-Fargallo, Towards the quantification of energy demand and consumption through the adaptive comfort approach in mixed mode office buildings considering climate change, Energy Build. 187 (2019) 173–185. https://doi.org/10.1016/j.enbuild.2019.02.002.
- [157] J. Younes, M. Chen, K. Ghali, R. Kosonen, A.K. Melikov, A.V. Farahani, S. Kilpeläinen, N. Ghaddar, Enhancing sustainability and resilience of elderly dwellings: Optimized refurbishing parameters and air conditioning operation, Energy Build. 289 (2023) 113065. https://doi.org/10.1016/j.enbuild.2023.113065.
- [158] P. Shakya, G. Ng, X. Zhou, Y.W. Wong, S. Dubey, S. Qian, Thermal comfort and energy analysis of a hybrid cooling system by coupling natural ventilation with radiant and indirect evaporative cooling, Energies (Basel). 14 (2021). https://doi.org/10.3390/en14227825.
- [159] J. Saif, A. Wright, S. Khattak, K. Elfadli, Keeping cool in the desert: Using wind catchers for improved thermal comfort and indoor air quality at half the energy, Buildings. 11 (2021). https://doi.org/10.3390/buildings11030100.
- [160] A. Mostafaeipour, H. Goudarzi, M. Khanmohammadi, M. Jahangiri, A. Sedaghat, H. Norouzianpour, S. Chowdhury, K. Techato, A. Issakhov, K. Almutairi, S.J. Hosseini Dehshiri, Techno-economic analysis and energy performance of a geothermal earth-to-air heat exchanger (EAHE) system in residential buildings: A case study, Energy Sci Eng. 9 (2021) 1807–1825. https://doi.org/10.1002/ese3.952.
- [161] S. Omrani, S. Matour, K. Bamdad, N. Izadyar, Ceiling fans as ventilation assisting devices in buildings: A critical review, Build Environ. 201 (2021) 108010. https://doi.org/10.1016/j.buildenv.2021.108010.
- [162] M. Veselý, W. Zeiler, Personalized conditioning and its impact on thermal comfort and energy performance A review, Renewable and Sustainable Energy Reviews. 34 (2014) 401–408. https://doi.org/10.1016/j.rser.2014.03.024.
- [163] K. Bamdad, S. Matour, N. Izadyar, S. Omrani, Impact of climate change on energy saving potentials of natural ventilation and ceiling fans in mixed-mode buildings, Build Environ. 209 (2022). https://doi.org/10.1016/j.buildenv.2021.108662.
- [164] M. André, A. Kamimura, M. Bavaresco, R.F. Giaretta, M. Fossati, R. Lamberts, Achieving mid-rise NZEB offices in Brazilian urban centres: A control strategy with desk fans and extension of set point temperature, Energy Build. 259 (2022). https://doi.org/10.1016/j.enbuild.2022.111911.
- [165] D. Bienvenido-Huertas, D. Sánchez-García, C. Rubio-Bellido, J. Solís-Guzmán, Using adaptive strategies of natural ventilation with tolerances applied to the upper limit to improve social dwellings' thermal comfort in current and future scenarios, Sci Technol Built Environ. 28 (2022) 527–546. https://doi.org/10.1080/23744731.2022.2040884.
- [166] N.R. Martins, P.J. Bourne-Webb, An assessment of the potential effect of climate change on renewable hybrid heating and cooling systems, Journal of Building Engineering. 64 (2023). https://doi.org/10.1016/j.jobe.2022.105598.
- [167] Y. Kim, Y. Choi, S.K. Min, Future changes in heat wave characteristics and their impacts on the electricity demand in South Korea, Weather Clim Extrem. 37 (2022) 100485. https://doi.org/10.1016/j.wace.2022.100485.
- [168] L. Borghero, E. Clèries, T. Péan, J. Ortiz, J. Salom, Comparing cooling strategies to assess thermal comfort resilience of residential buildings in Barcelona for present and future heatwaves, Build Environ. 231 (2023). https://doi.org/10.1016/j.buildenv.2023.110043.
- [169] A. Figueroa-Lopez, A. Arias, X. Oregi, I. Rodríguez, Evaluation of passive strategies, natural ventilation and shading systems, to reduce overheating risk in a passive house tower in the north of Spain during the warm season, Journal of Building Engineering. 43 (2021). https://doi.org/10.1016/j.jobe.2021.102607.
- [170] B. Widera, Comparative analysis of user comfort and thermal performance of six types of vernacular dwellings as the first step towards climate resilient, sustainable and bioclimatic architecture in western sub-Saharan Africa, Renewable and Sustainable Energy Reviews. 140 (2021). https://doi.org/10.1016/j.rser.2021.110736.
- [171] F.M. Baba, H. Ge, L.L. Wang, R. Zmeureanu, Do high energy-efficient buildings increase overheating risk in cold climates? Causes and mitigation measures required under recent and future climates, Build Environ. 219 (2022). https://doi.org/10.1016/j.buildenv.2022.109230.
- [172] C. Du, W. Yu, Y. Ma, Q. Cai, B. Li, N. Li, W. Wang, R. Yao, A holistic investigation into the seasonal and temporal variations of window opening behavior in residential buildings in Chongqing, China, Energy Build. 231 (2021). https://doi.org/10.1016/j.enbuild.2020.110522.
- [173] E. Belias, D. Licina, Influence of outdoor air pollution on European residential ventilative cooling potential, Energy Build. 289 (2023) 113044. https://doi.org/10.1016/j.enbuild.2023.113044.
- [174] M. Barclay, J. Kang, S. Sharples, Combining noise mapping and ventilation performance for non-domestic buildings in an urban area, Build Environ. 52 (2012) 68–76. https://doi.org/10.1016/j.buildenv.2011.12.015.
- [175] R. Consentino Kronka Mulfarth, J. Carla Soares Gonçalves, R. Loureiro Xavier Nascimento Michalski Michalski, A. Rodrigues Prata Shimomura, B. Nascimento e Souza, G. Reis Muri Cunha, M. Pereira Marcondes-Cavaleri, M. Alvarez Gomez Monroy, The environmental response of office buildings from the Brazilian bioclimatic modernism in São Paulo: the case study of Complexo Conjunto Nacional CCN (1962), Solar Energy. 220 (2021) 617–634. https://doi.org/10.1016/j.solener.2021.02.059.

- [176] C.M. Calama-González, Á.L. León-Rodríguez, R. Suárez, Climate change mitigation: thermal comfort improvement in Mediterranean social dwellings through dynamic test cells modelling, International Journal of Energy and Environmental Engineering. 14 (2023) 121–134. https://doi.org/10.1007/s40095-022-00498-1.
- [177] S. Truchet, A. Jay, E. Wurtz, J. Anger, A. Brun, P. Bernaud, Impact of thermal inertia coupled to natural night ventilation. A case study for a high performance building in continental climate, International Journal of Ventilation. (2023). https://doi.org/10.1080/14733315.2023.2188346.
- [178] Z. Wang, B. Cao, Y. Zhu, Questionnaire survey and field investigation on sleep thermal comfort and behavioral adjustments in bedrooms of Chinese residents, Energy Build. 253 (2021) 111462. https://doi.org/10.1016/j.enbuild.2021.111462.
- [179] Y. Liu, Y. Gao, C. Zhuang, D. Shi, Y. Xu, J. Guan, Y. Di, Optimization of top-floor rooms coupling cool roofs, natural ventilation and solar shading for residential buildings in hot-summer and warm-winter zones, Journal of Building Engineering. 66 (2023). https://doi.org/10.1016/j.jobe.2023.105933.
- [180] ASHRAE, Standard 62.1-2022. Ventilation and Acceptable Indoor Air Quality, (n.d.).
- [181] M. Shrestha, H.B. Rijal, G. Kayo, M. Shukuya, An investigation on CO2 concentration based on field survey and simulation in naturally ventilated Nepalese school buildings during summer, Build Environ. 207 (2022). https://doi.org/10.1016/j.buildenv.2021.108405.
- [182] M. Habil, A. Taneja, Children's exposure to indoor particulate matter in naturally ventilated schools in India, Indoor and Built Environment. 20 (2011) 430–448. https://doi.org/10.1177/1420326X11409455.
- [183] M.A. Campano-Laborda, S. Domínguez-Amarillo, J. Fernández-Agüera, I. Acosta, Indoor comfort and symptomatology in non-university educational buildings: Occupants' perception, Atmosphere (Basel). 11 (2020). https://doi.org/10.3390/atmos11040357.
- [184] K. Huang, J. Song, G. Feng, Q. Chang, B. Jiang, J. Wang, W. Sun, H. Li, J. Wang, X. Fang, Indoor air quality analysis of residential buildings in northeast China based on field measurements and longtime monitoring, Build Environ. 144 (2018) 171–183. https://doi.org/10.1016/j.buildenv.2018.08.022.
- [185] H.K. Abdullah, H.Z. Alibaba, Window design of naturally ventilated offices in the mediterranean climate in terms of CO2 and thermal comfort performance, Sustainability (Switzerland). 12 (2020). https://doi.org/10.3390/su12020473.
- [186] E.Z.E. Conceição, J.M.M. Gomes, N.H. Antão, M.M.J.R. Lúcio, Application of a developed adaptive model in the evaluation of thermal comfort in ventilated kindergarten occupied spaces, Build Environ. 50 (2012) 190–201. https://doi.org/10.1016/j.buildenv.2011.10.013.
- [187] W. Yang, J. Sohn, J. Kim, B. Son, J. Park, Indoor air quality investigation according to age of the school buildings in Korea, J Environ Manage. 90 (2009) 348–354. https://doi.org/10.1016/j.jenvman.2007.10.003.
- [188] M.M.M. Abdel-Salam, Seasonal variation in indoor concentrations of air pollutants in residential buildings, J Air Waste Manage Assoc. 71 (2021) 761–777. https://doi.org/10.1080/10962247.2021.1895367.
- [189] M.M.M. Abdel-Salam, Outdoor and indoor factors influencing particulate matter and carbon dioxide levels in naturally ventilated urban homes, J Air Waste Manage Assoc. 71 (2021) 60–69. https://doi.org/10.1080/10962247.2020.1834009.
- [190] N. Szczepanik-Scislo, L. Scislo, Comparison of cfd and multizone modeling from contaminant migration from a household gas furnace, Atmosphere (Basel). 12 (2021). https://doi.org/10.3390/ATMOS12010079.
- [191] V.S. Chithra, S.M. Shiva Nagendra, Indoor air quality investigations in a naturally ventilated school building located close to an urban roadway in Chennai, India, Build Environ. 54 (2012) 159–167. https://doi.org/10.1016/j.buildenv.2012.01.016.
- [192] Sources of Indoor Particulate Matter (PM). Available online: https://www.epa.gov/indoor-air-quality-iaq/sourcesindoor-particulate-matter-pm. (Accessed 16 June 2023)., (n.d.).
- [193] V.S. Chithra, S.M. Shiva Nagendra, Seasonal trends of indoor particulate matter concentrations in a naturally ventilated school building, 22nd International Conference on Modelling, Monitoring and Management of Air Pollution, AIR 2014. 183 (2014) 341–351. https://doi.org/10.2495/AIR140281.
- [194] S. Mentese, A.Y. Rad, M. Arisoy, G. Güllü, Multiple comparisons of organic, microbial, and fine particulate pollutants in typical indoor environments: Diurnal and seasonal variations, J Air Waste Manage Assoc. 62 (2012) 1380–1393. https://doi.org/10.1080/10962247.2012.714717.
- [195] B. Oroji, A. Sadighzadeh, E. Solgi, M.S. Oliaei, Impact of air quality on students' behavior in the Educational Centers, Air Qual Atmos Health. 14 (2021) 793–806. https://doi.org/10.1007/s11869-021-00979-y.
- [196] V. Sahu, B.R. Gurjar, Spatial and seasonal variation of air quality in different microenvironments of a technical university in India, Build Environ. 185 (2020). https://doi.org/10.1016/j.buildenv.2020.107310.
- [197] E. Belias, D. Licina, Influence of outdoor air pollution on European residential ventilative cooling potential, Energy Build. 289 (2023) 113044. https://doi.org/10.1016/j.enbuild.2023.113044.
- [198] C. Li, L. Bai, Z. He, X. Liu, X. Xu, The effect of air purifiers on the reduction in indoor PM2.5 concentrations and population health improvement, Sustain Cities Soc. 75 (2021). https://doi.org/10.1016/j.scs.2021.103298.
- [199] S. Stamp, E. Burman, C. Shrubsole, L. Chatzidiakou, D. Mumovic, M. Davies, Long-term, continuous air quality monitoring in a cross-sectional study of three UK non-domestic buildings, Build Environ. 180 (2020). https://doi.org/10.1016/j.buildenv.2020.107071.
- [200] A. Fonseca, I. Abreu, M.J. Guerreiro, C. Abreu, R. Silva, N. Barros, Indoor air quality and sustainability management-Case study in three Portuguese healthcare units, Sustainability (Switzerland). 11 (2019). https://doi.org/10.3390/su11010101.

- [201] A. Núñez, A.M. García, The aerobiome in a hospital environment: Characterization, seasonal tendencies and the effect of window opening ventilation, Build Environ. 230 (2023). https://doi.org/10.1016/j.buildenv.2023.110024.
- [202] A. Mendes, C. Pereira, D. Mendes, L. Aguiar, P. Neves, S. Silva, S. Batterman, J.P. Teixeira, Indoor air quality and thermal comfort - Results of a pilot study in elderly care centers in portugal, Journal of Toxicology and Environmental Health - Part A: Current Issues. 76 (2013) 333–344. https://doi.org/10.1080/15287394.2013.757213.
- [203] A. Asif, M. Zeeshan, Comparative analysis of indoor air quality in offices with different ventilation mechanisms and simulation of ventilation process utilizing system dynamics tool, Journal of Building Engineering. 72 (2023). https://doi.org/10.1016/j.jobe.2023.106687.
- [204] N.R. Kapoor, A. Kumar, A. Kumar, D.A. Zebari, K. Kumar, M.A. Mohammed, A.S. Al-Waisy, M.A. Albahar, Event-Specific Transmission Forecasting of SARS-CoV-2 in a Mixed-Mode Ventilated Office Room Using an ANN, Int J Environ Res Public Health. 19 (2022). https://doi.org/10.3390/ijerph192416862.
- [205] H.C. Burridge, S. Bontitsopoulos, C. Brown, H. Carter, K. Roberts, C. Vouriot, D. Weston, M. Mon-Williams, N. Williams, C. Noakes, Variations in classroom ventilation during the COVID-19 pandemic: Insights from monitoring 36 naturally ventilated classrooms in the UK during 2021, Journal of Building Engineering. 63 (2023). https://doi.org/10.1016/j.jobe.2022.105459.
- [206] A.J. Aguilar, M.L. de la Hoz-Torres, N. Costa, P. Arezes, M.D. Martínez-Aires, D.P. Ruiz, Assessment of ventilation rates inside educational buildings in Southwestern Europe: Analysis of implemented strategic measures, Journal of Building Engineering. 51 (2022) 104204. https://doi.org/10.1016/J.JOBE.2022.104204.
- [207] F.F. Duill, F. Schulz, A. Jain, L. Krieger, B. van Wachem, F. Beyrau, The impact of large mobile air purifiers on aerosol concentration in classrooms and the reduction of airborne transmission of sars-cov-2, Int J Environ Res Public Health. 18 (2021). https://doi.org/10.3390/ijerph182111523.
- [208] E.S. Wijaya, Natural ventilation optimization study in mechanically ventilated studio apartment room in surabaya, Journal of Applied Science and Engineering (Taiwan). 25 (2022) 141–149. https://doi.org/10.6180/jase.202202_25(1).0014.
- [209] F. Xu, Z. Gao, Study on indoor air quality and fresh air energy consumption under different ventilation modes in 24hour occupied bedrooms in Nanjing, using Modelica-based simulation, Energy Build. 257 (2022). https://doi.org/10.1016/j.enbuild.2021.111805.
- [210] L.J.R. Nunes, A. Curado, S.I. Lopes, Understanding Seasonal Indoor Radon Variability from Data Collected with a LoRa-Enabled IoT Edge Device, Applied Sciences (Switzerland). 13 (2023). https://doi.org/10.3390/app13084735.
- [211] D. Al-Azmi, A.I. Abu-Shady, A.M. Sayed, Y. Al-Zayed, Indoor radon in Kuwait, Health Phys. 94 (2008) 49–56. https://doi.org/10.1097/01.HP.0000281177.42452.8f.
- [212] T. Deng, X. Shen, X. Cheng, J. Liu, Investigation of window-opening behaviour and indoor air quality in dwellings situated in the temperate zone in China, Indoor and Built Environment. 30 (2021) 938–956. https://doi.org/10.1177/1420326X20924746.
- [213] M. Kolokotroni, Y.T. Ge, D. Katsoulas, Monitoring and modelling indoor air quality and ventilation in classrooms within a purpose-designed naturally ventilated school, Indoor and Built Environment. 11 (2002) 316–326. https://doi.org/10.1159/000068232.
- [214] G.Y. Yun, H. Kim, J.T. Kim, Thermal and non-thermal stimuli for the use of windows in offices, Indoor and Built Environment. 21 (2012) 109–121. https://doi.org/10.1177/1420326X11420012.
- [215] G.S. Graudenz, C.H. Oliveira, A. Tribess, C. Mendes Jr., M.R.D.O. Latorre, J. Kalil, Association of air-conditioning with respiratory symptoms in office workers in tropical climate, Indoor Air. 15 (2005) 62–66. https://doi.org/10.1111/j.1600-0668.2004.00324.x.
- [216] C. Du, W. Yu, Y. Ma, Q. Cai, B. Li, N. Li, W. Wang, R. Yao, A holistic investigation into the seasonal and temporal variations of window opening behavior in residential buildings in Chongqing, China, Energy Build. 231 (2021). https://doi.org/10.1016/j.enbuild.2020.110522.
- [217] M. Baborska-Narożny, M. Kostka, Seasonal Air Quality in Bedrooms with Natural, Mechanical or Hybrid Ventilation Systems and Varied Window Opening Behavior-Field Measurement Results, Energies (Basel). 15 (2022). https://doi.org/10.3390/en15249328.
- [218] M.F. Mohamed, M. Behnia, S. King, D. Prasad, The potential of natural ventilation in single-sided ventilated apartment to improve indoor thermal comfort and air quality, in: ASME 2011 5th International Conference on Energy Sustainability, ES 2011, Faculty of Engineering and Built Environment, Universiti Kebangsaan Malaysia Bangi, Selangor, Malaysia, 2011: pp. 71–78. https://doi.org/10.1115/ES2011-54129.
- [219] J. Zong, W. He, L. Wang, H. Chai, Y. Zhao, J. Liu, Experimental study on indoor environment quality in a naturally ventilated classroom of a university using natural ventilation and ventilation fan, in: Q. W. (Ed.), 2nd International Conference on Civil Architecture and Energy Science, CAES 2020, EDP Sciences, School of Thermal Engineering, Shandong Jianzhu University, Jinan, 250101, China, 2020. https://doi.org/10.1051/e3sconf/202016504082.
- [220] J.S. Carlos, Improving indoor air quality in school buildings, enhancing comfort and reducing heating energy consumption through the use of a passive system, in: Natural Ventilation: Strategies, Health Implications and Impacts on the Environment, Nova Science Publishers, Inc., C-MADE, Centre of Materials and Building Technologies, University of Beira Interior, Covilhã, Portugal, 2015: pp. 117–146. https://www.scopus.com/inward/record.uri?eid=2-s2.0-

84956818175&partnerID=40&md5=1eceecd3edac4b35b7cb0701a2e2112c.

[221] A. Mavrogianni, D. Mumovic, Onthe use of windcatchers in schools: Climate change, occupancy patterns, and adaptation strategies, Indoor and Built Environment. 19 (2010) 340–354. https://doi.org/10.1177/1420326X09341507.

- [222] A.A. Elmualim, Failure of a control strategy for a hybrid air-conditioning and wind catchers/towers system at Bluewater shopping malls in Kent, UK, Facilities. 24 (2006) 399–411. https://doi.org/10.1108/02632770610684882.
- [223] R.T. Hellwig, M. Sedlmeier, C. Tanzer, Prospects of improving the indoor air quality in classrooms by reactivating historic ventilation stacks, International Journal of Ventilation. 14 (2015) 141–152. https://doi.org/10.1080/14733315.2015.11684076.
- [224] E.H. Moghaddam, S. Amindeldar, A. Besharatizadeh, New approach to natural ventilation in public buildings inspired by Iranian's traditional windcatcher, in: 2011 International Conference on Green Buildings and Sustainable Cities, GBSC 2011, Elsevier Ltd, University of Tehran, Inqelab, Tehran, Iran, 2011: pp. 42–52. https://doi.org/10.1016/j.proeng.2011.11.1985.
- [225] B.M. Jones, R. Kirby, Indoor air quality in U.K. school classrooms ventilated by natural ventilation windcatchers, International Journal of Ventilation. 10 (2012) 323–338. https://doi.org/10.1080/14733315.2012.11683959.
- [226] M. Singh, R. Sharston, A literature review of building energy simulation and computational fluid dynamics cosimulation strategies and its implications on the accuracy of energy predictions, Building Services Engineering Research and Technology. 43 (2021) 113–138. https://doi.org/10.1177/01436244211020465.
- [227] S. Subhashini, K. Thirumaran, CFD simulations for examining natural ventilation in the learning spaces of an educational building with courtyards in Madurai, Building Services Engineering Research and Technology. 41 (2020) 466–479. https://doi.org/10.1177/0143624419878798.
- [228] C. Yuan, R. Zhu, S. Tong, S. Mei, W. Zhu, Impact of anthropogenic heat from air-conditioning on air temperature of naturally ventilated apartments at high-density tropical cities, Energy Build. 268 (2022). https://doi.org/10.1016/j.enbuild.2022.112171.
- [229] M. Xie, Y. Wang, Z. Liu, G. Zhang, Effect of the location pattern of rural residential buildings on natural ventilation in mountainous terrain of central China, J Clean Prod. 340 (2022). https://doi.org/10.1016/j.jclepro.2022.130837.
- [230] A. Soto, P. Martínez, V.M. Soto, P.J. Martínez, Analysis of the performance of a passive downdraught evaporative cooling system driven by solar chimneys in a residential building by using an experimentally validated trnsys model, Energies (Basel). 14 (2021). https://doi.org/10.3390/en14123486.
- [231] W. Lv, C. Shen, X. Li, Energy efficiency of an air conditioning system coupled with a pipe-embedded wall and mechanical ventilation, Journal of Building Engineering. 15 (2018) 229–235. https://doi.org/10.1016/j.jobe.2017.11.010.
- [232] W.A.Y. Mousa, W. Lang, T. Auer, Assessment of the impact of window screens on indoor thermal comfort and energy efficiency in a naturally ventilated courtyard house, Archit Sci Rev. 60 (2017) 382–394. https://doi.org/10.1080/00038628.2017.1329134.
- [233] W. Zhong, W. Xiao, T. Zhang, Numerical Investigations on Natural Ventilation in Atria of China's Southern Yangtze Vernacular Dwellings, Sustain Cities Soc. 89 (2023). https://doi.org/10.1016/j.scs.2022.104341.
- [234] E. Spentzou, M.J. Cook, S. Emmitt, Modelling natural ventilation for summer thermal comfort in Mediterranean dwellings, International Journal of Ventilation. 18 (2019) 28–45. https://doi.org/10.1080/14733315.2017.1302658.
- [235] C.-M. Chiang, C.-M. Lai, The Influence of Ventilation Design Mixes on the Thermal Environment in Elementary School Classrooms, Indoor and Built Environment. 18 (2009) 346–359. https://doi.org/10.1177/1420326X09337047.
- [236] Z.A. Adamu, A. Price, The design and simulation of natural personalised ventilation (NPV) system for multi-bed hospital wards, Buildings. 5 (2015) 381–404. https://doi.org/10.3390/buildings5020381.
- [237] Y. Liu, Y. Gao, C. Zhuang, D. Shi, Y. Xu, J. Guan, Y. Di, Optimization of top-floor rooms coupling cool roofs, natural ventilation and solar shading for residential buildings in hot-summer and warm-winter zones, Journal of Building Engineering. 66 (2023). https://doi.org/10.1016/j.jobe.2023.105933.
- [238] A.N. Zoure, P.V. Genovese, Implementing natural ventilation and daylighting strategies for thermal comfort and energy efficiency in office buildings in Burkina Faso, Energy Reports. 9 (2023) 3319–3342. https://doi.org/10.1016/j.egyr.2023.02.017.
- [239] K. Grygierek, J. Ferdyn-Grygierek, Design of Ventilation Systems in a Single-Family House in Terms of Heating Demand and Indoor Environment Quality, Energies (Basel). 15 (2022). https://doi.org/10.3390/en15228456.
- [240] Z. Zeng, X. Li, C. Li, Y. Zhu, Modeling ventilation in naturally ventilated double-skin façade with a venetian blind, Build Environ. 57 (2012) 1–6. https://doi.org/10.1016/j.buildenv.2012.04.007.
- [241] S. Latif, B. Hamzah, R. Rahim, R. Mulyadi, A.E.E. Putra, Study of the heat ventilation with inclined chimney in the attic, International Journal of GEOMATE. 17 (2019) 178–184. https://doi.org/10.21660/2019.64.15375.
- [242] H. Laloui, N.H. Abdul Majid, A.N.Z. Sanusi, The impacts of introducing voids combinations on indoor ventilation performance in high-rise residential buildings, Engineering, Construction and Architectural Management. 29 (2022) 2736–2759. https://doi.org/10.1108/ECAM-09-2020-0752.
- [243] Y. Hou, A. Li, S. Mei, Learning from Chinese traditional architecture: field test and CFD modelling of ventilation enhancement techniques in southern Chinese houses, International Journal of Ventilation. 21 (2022) 1–18. https://doi.org/10.1080/14733315.2021.1876406.
- [244] H.D. Mohamadabadi, A.H. Ghanbaran, A.A. Dehghan, The study of air distribution quality in the summer section of Iranian dry climate houses equipped with wind tower, Sustain Cities Soc. 86 (2022). https://doi.org/10.1016/j.scs.2022.104095.
- [245] S. Mohamed, H. Al-Khatri, J. Calautit, S. Omer, S. Riffat, The impact of a passive wall combining natural ventilation and evaporative cooling on schools' thermal conditions in a hot climate, Journal of Building Engineering. 44 (2021). https://doi.org/10.1016/j.jobe.2021.102624.

- [246] V. Vitale, G. Salerno, A numerical prediction of the passive cooling effects on thermal comfort for a historical building in Rome, Energy Build. 157 (2017) 1–10. https://doi.org/10.1016/j.enbuild.2017.06.049.
- [247] D. Prakash, P. Ravikumar, Analysis of thermal comfort in a residential room with multiple vents: A case study by numerical simulation technique, Research Journal of Applied Sciences, Engineering and Technology. 6 (2013) 4164–4173. https://doi.org/10.19026/rjaset.6.3527.
- [248] D. Li, Y. Zheng, C. Liu, H. Qi, X. Liu, Numerical analysis on thermal performance of naturally ventilated roofs with different influencing parameters, Sustain Cities Soc. 22 (2016) 86–93. https://doi.org/10.1016/j.scs.2016.02.004.
- [249] A.M. Nugroho, M.H. Ahmad, D.R. Ossen, A Preliminary Study of Thermal Comfort in Malaysia's Single Storey Terraced Houses, Journal of Asian Architecture and Building Engineering. 6 (2007) 175–182. https://doi.org/10.3130/jaabe.6.175.
- [250] S. Gunasagaran, E.S. Saw, T. Mari, S. Srirangam, V. Ng, Courtyard configuration to optimize shading, daylight and ventilation in a tropical terrace house using simulation, International Journal of Architectural Research: Archnet-IJAR. 17 (2023) 109–123. https://doi.org/10.1108/ARCH-12-2021-0354.
- [251] Q. Sun, Z. Luo, L. Bai, The Impact of Internal Courtyard Configuration on Thermal Performance of Long Strip Houses, Buildings. 13 (2023). https://doi.org/10.3390/buildings13020371.
- [252] S. Hao, C. Yu, Y. Xu, Y. Song, The effects of courtyards on the thermal performance of a vernacular house in a hotsummer and cold-winter climate, Energies (Basel). 12 (2019). https://doi.org/10.3390/en12061042.
- [253] W.A. Yousef Mousa, W. Lang, T. Auer, Numerical assessment of the efficiency of fenestration system and natural ventilation mechanisms in a courtyard house in hot climate, Build Simul. 10 (2017) 737–754. https://doi.org/10.1007/s12273-017-0357-0.
- [254] S. Liu, C. Huang, Optimization of natural ventilation of "yinzi" dwellings in western hunan based on orthogonal experiment and CFD, Buildings. 6 (2016). https://doi.org/10.3390/buildings6030025.
- [255] X. Du, R. Bokel, A. van den Dobbelsteen, Building microclimate and summer thermal comfort in free-running buildings with diverse spaces: A Chinese vernacular house case, Build Environ. 82 (2014) 215–227. https://doi.org/10.1016/j.buildenv.2014.08.022.
- [256] J.K. Calautit, B.R. Hughes, S.A. Ghani, A numerical investigation into the feasibility of integrating green building technologies into row houses in the Middle East, Archit Sci Rev. 56 (2013) 279–296. https://doi.org/10.1080/00038628.2012.686433.
- [257] A. Balabel, M. Faizan, A. Alzaed, Towards a Computational Fluid Dynamics-Based Fuzzy Logic Controller of the Optimum Windcatcher Internal Design for Efficient Natural Ventilation in Buildings, Math Probl Eng. 2021 (2021). https://doi.org/10.1155/2021/9936178.
- [258] T. Kobayashi, T. Chikamoto, K. Osada, Evaluation of ventilation performance of monitor roof in residential area based on simplified estimation and CFD analysis, Build Environ. 63 (2013) 20–30. https://doi.org/10.1016/j.buildenv.2013.01.018.
- [259] M. Lança, P.J. Coelho, J. Viegas, Numerical simulation of a night cooling strategy in an office room, Energy Build. 252 (2021). https://doi.org/10.1016/j.enbuild.2021.111359.
- [260] S. Suhendri, M. Hu, Y. Su, J. Darkwa, S. Riffat, Parametric study of a novel combination of solar chimney and radiative cooling cavity for natural ventilation enhancement in residential buildings, Build Environ. 225 (2022). https://doi.org/10.1016/j.buildenv.2022.109648.
- [261] Z. Moghtader Gilvaei, A. Haghighi Poshtiri, A. Mirzazade Akbarpoor, A novel passive system for providing natural ventilation and passive cooling: Evaluating thermal comfort and building energy, Renew Energy. 198 (2022) 463– 483. https://doi.org/10.1016/j.renene.2022.07.151.
- [262] L. Moosavi, M. Zandi, M. Bidi, E. Behroozizade, I. Kazemi, New design for solar chimney with integrated windcatcher for space cooling and ventilation, Build Environ. 181 (2020). https://doi.org/10.1016/j.buildenv.2020.106785.
- [263] C. Piselli, M. Prabhakar, A. de Gracia, M. Saffari, A.L. Pisello, L.F. Cabeza, Optimal control of natural ventilation as passive cooling strategy for improving the energy performance of building envelope with PCM integration, Renew Energy. 162 (2020) 171–181. https://doi.org/10.1016/j.renene.2020.07.043.
- [264] Ö. Duran, K.J. Lomas, Retrofitting post-war office buildings: Interventions for energy efficiency, improved comfort, productivity and cost reduction, Journal of Building Engineering. 42 (2021). https://doi.org/10.1016/j.jobe.2021.102746.
- [265] M. André, A. Kamimura, M. Bavaresco, R.F. Giaretta, M. Fossati, R. Lamberts, Achieving mid-rise NZEB offices in Brazilian urban centres: A control strategy with desk fans and extension of set point temperature, Energy Build. 259 (2022). https://doi.org/10.1016/j.enbuild.2022.111911.
- [266] D. Bienvenido-Huertas, D. Sánchez-García, C. Rubio-Bellido, Analysing natural ventilation to reduce the cooling energy consumption and the fuel poverty of social dwellings in coastal zones, Appl Energy. 279 (2020). https://doi.org/10.1016/j.apenergy.2020.115845.
- [267] M. Prabhakar, M. Saffari, A. de Gracia, L.F. Cabeza, Improving the energy efficiency of passive PCM system using controlled natural ventilation, Energy Build. 228 (2020). https://doi.org/10.1016/j.enbuild.2020.110483.
- [268] R. Yao, V. Costanzo, X. Li, Q. Zhang, B. Li, The effect of passive measures on thermal comfort and energy conservation. A case study of the hot summer and cold winter climate in the Yangtze River region, Journal of Building Engineering. 15 (2018) 298–310. https://doi.org/10.1016/j.jobe.2017.11.012.
- [269] G. Chiesa, N. Huberman, D. Pearlmutter, Geo-climatic potential of direct evaporative cooling in the Mediterranean Region: A comparison of key performance indicators, Build Environ. 151 (2019) 318–337. https://doi.org/10.1016/j.buildenv.2019.01.059.

- [270] M.K. Nematchoua, J.C. V Noelson, I. Saadi, H. Kenfack, A.-Z.F.R. Andrianaharinjaka, D.F. Ngoumdoum, J.B. Sela, S. Reiter, Application of phase change materials, thermal insulation, and external shading for thermal comfort improvement and cooling energy demand reduction in an office building under different coastal tropical climates, Solar Energy. 207 (2020) 458–470. https://doi.org/10.1016/j.solener.2020.06.110.
- [271] Z.A. Elhassan, Energy consumption performance using natural ventilation in dwelling design and CFD simulation in a hot dry climate: A case study in Sudan, Front Built Environ. 9 (2023). https://doi.org/10.3389/fbuil.2023.1145747.
- [272] V.L. Visakha, T. Srinivas, EVALUATING THE CLIMATIC POTENTIAL OF PASSIVE STRATEGIES FOR RESIDENCES IN FOUR CITIES UNDER THE WARM-HUMID CLIMATE OF SOUTH INDIA, Journal of Green Building. 18 (2023) 147–168. https://doi.org/10.3992/jgb.18.1.147.
- [273] J. Daaboul, K. Ghali, N. Ghaddar, Mixed-mode ventilation and air conditioning as alternative for energy savings: a case study in Beirut current and future climate, Energy Effic. 11 (2018) 13–30. https://doi.org/10.1007/s12053-017-9546-z.
- [274] C. Schünemann, D. Schiela, R. Ortlepp, How window ventilation behaviour affects the heat resilience in multiresidential buildings, Build Environ. 202 (2021). https://doi.org/10.1016/j.buildenv.2021.107987.
- [275] R. Simson, J. Kurnitski, K. Kuusk, Experimental validation of simulation and measurement-based overheating assessment approaches for residential buildings, Archit Sci Rev. 60 (2017) 192–204. https://doi.org/10.1080/00038628.2017.1300130.
- [276] M. Liu, K.B. Wittchen, P.K. Heiselberg, Verification of a simplified method for intelligent glazed façade design under different control strategies in a full-scale façade test facility - Preliminary results of a south facing single zone experiment for a limited summer period, Build Environ. 82 (2014) 400–407. https://doi.org/10.1016/j.buildenv.2014.09.009.
- [277] M. Macias, J.A. Gaona, J.M. Luxan, G. Gomez, Low cost passive cooling system for social housing in dry hot climate, Energy Build. 41 (2009) 915–921. https://doi.org/10.1016/j.enbuild.2009.03.013.
- [278] F.-Z. El-Bichri, I. Sobhy, I. Bouchefra, B. Benhamou, H. Chehouani, M.O. Mghazli, Assessment of the impact of construction materials on the building's thermal behaviour and indoor thermal comfort in a hot and semi-arid climate, Advances in Building Energy Research. 16 (2022) 711–735. https://doi.org/10.1080/17512549.2022.2096692.
- [279] W. Liping, W.N. Hien, The impacts of ventilation strategies and facade on indoor thermal environment for naturally ventilated residential buildings in Singapore, Build Environ. 42 (2007) 4006–4015. https://doi.org/10.1016/j.buildenv.2006.06.027.
- [280] C. Simonson, Energy consumption and ventilation performance of a naturally ventilated ecological house in a cold climate, Energy Build. 37 (2005) 23–35. https://doi.org/10.1016/j.enbuild.2004.04.006.
- [281] E. Gratia, A. De Herde, Natural ventilation in a double-skin facade, Energy Build. 36 (2004) 137–146. https://doi.org/10.1016/j.enbuild.2003.10.008.
- [282] R.G. Ogden, C.C. Kendrick, N.S.R. Walliman, Modelling of enhanced passive and conventional cooling systems, Building Research and Information. 32 (2004) 17–26. https://doi.org/10.1080/09613210310000100788.
- [283] N.H. Wong, S. Li, A study of the effectiveness of passive climate control in naturally ventilated residential buildings in Singapore, Build Environ. 42 (2007) 1395–1405. https://doi.org/10.1016/j.buildenv.2005.11.032.
- [284] N.H. Wong, S. Li, Determination of acceptable u-values for naturally ventilated residential building façades in Singapore, Archit Sci Rev. 49 (2006) 156–161. https://doi.org/10.3763/asre.2006.4921.
- [285] J.S. Carlos, H. Corvacho, P.D. Silva, J.P. Castro-Gomes, Modelling and simulation of a ventilated double window, Appl Therm Eng. 31 (2011) 93–102. https://doi.org/10.1016/j.applthermaleng.2010.08.021.
- [286] N.A.M. Al-Tamimi, S.F. Syed Fadzil, Thermal performance analysis for ventilated and unventilated glazed rooms in Malaysia (comparing simulated and field data), Indoor and Built Environment. 20 (2011) 534–542. https://doi.org/10.1177/1420326X11411235.
- [287] Santy, H. Matsumoto, K. Tsuzuki, L. Susanti, Bioclimatic analysis in pre-design stage of passive house in Indonesia, Buildings. 7 (2017). https://doi.org/10.3390/buildings7010024.
- [288] X. Juan, L. Ziliang, G. Weijun, Y. Mengsheng, S. Menglong, The comparative study on the climate adaptability based on indoor physical environment of traditional dwelling in Qinba mountainous areas, China, Energy Build. 197 (2019) 140–155. https://doi.org/10.1016/j.enbuild.2019.05.045.
- [289] N. Areemit, Y. Sakamoto, Numerical and experimental analysis of a passive room-dehumidifying system using the sorption property of a wooden attic space, Energy Build. 39 (2007) 317–327. https://doi.org/10.1016/j.enbuild.2006.07.007.
- [290] J.-M. Alessandrini, J. Ribéron, D. Da Silva, Will naturally ventilated dwellings remain safe during heatwaves?, Energy Build. 183 (2019) 408–417. https://doi.org/10.1016/j.enbuild.2018.10.033.
- [291] S. Coccolo, D. Pearlmutter, J. Kaempf, J.-L. Scartezzini, Thermal Comfort Maps to estimate the impact of urban greening on the outdoor human comfort, Urban For Urban Green. 35 (2018) 91–105. https://doi.org/10.1016/j.ufug.2018.08.007.
- [292] A. Brambilla, J. Bonvin, F. Flourentzou, T. Jusselme, On the influence of thermal mass and natural ventilation on overheating risk in offices, Buildings. 8 (2018). https://doi.org/10.3390/buildings8040047.
- [293] F. Flourentzou, S. Pantet, K. Ritz, Design and performance of controlled natural ventilation in school gymnasiums, International Journal of Ventilation. 16 (2017) 112–123. https://doi.org/10.1080/14733315.2016.1220202.
- [294] Y. Xiang, G. Zhou, Thermal performance of a window-based cooling unit using phase change materials combined with night ventilation, Energy Build. 108 (2015) 267–278. https://doi.org/10.1016/j.enbuild.2015.09.030.

- [295] X. Li, K. Huang, G. Feng, D. Jing, D. Liu, J. Li, Analysis on night ventilation effect of buildings with different energy consumption levels in Shenyang, in: L. A., O. T., K. R. (Eds.), 16th ROOMVENT Conference, ROOMVENT 2022, EDP Sciences, School of Municipal and Environmental Engineering, Shenyang Jianzhu University, Shenyang, 110168, China, 2022. https://doi.org/10.1051/e3sconf/202235601066.
- [296] Z. Huifen, Y. Fuhua, Z. Qian, Research on the Impact of Wind Angles on the Residential Building Energy Consumption, Math Probl Eng. 2014 (2014). https://doi.org/10.1155/2014/794650.
- [297] L. Ben-Alon, A.R. Rempel, Thermal comfort and passive survivability in earthen buildings, Build Environ. 238 (2023). https://doi.org/10.1016/j.buildenv.2023.110339.
- [298] J. Gamero-Salinas, A. Monge-Barrio, N. Kishnani, J. López-Fidalgo, A. Sánchez-Ostiz, Passive cooling design strategies as adaptation measures for lowering the indoor overheating risk in tropical climates, Energy Build. 252 (2021). https://doi.org/10.1016/j.enbuild.2021.111417.
- [299] K. Sun, W. Zhang, Z. Zeng, R. Levinson, M. Wei, T. Hong, Passive cooling designs to improve heat resilience of homes in underserved and vulnerable communities, Energy Build. 252 (2021). https://doi.org/10.1016/j.enbuild.2021.111383.
- [300] L. Scheuring, B. Weller, An investigation of ventilation control strategies for louver windows in different climate zones, International Journal of Ventilation. 20 (2021) 226–235. https://doi.org/10.1080/14733315.2020.1777018.
- [301] F. Ascione, N. Bianco, R.F. De Masi, F. de' Rossi, G.P. Vanoli, Energy refurbishment of existing buildings through the use of phase change materials: Energy savings and indoor comfort in the cooling season, Appl Energy. 113 (2014) 990–1007. https://doi.org/10.1016/j.apenergy.2013.08.045.
- [302] Y. Zou, J. Guo, D. Xia, S. Lou, Y. Huang, X. Yang, Z. Zhong, Quantitative analysis and enhancement on passive survivability of vernacular houses in the hot and humid region of China, Journal of Building Engineering. 71 (2023). https://doi.org/10.1016/j.jobe.2023.106431.
- [303] C. Liu, H. Mohammadpourkarbasi, S. Sharples, Evaluating the potential energy savings of retrofitting low-rise suburban dwellings towards the Passivhaus EnerPHit standard in a hot summer/cold winter region of China, Energy Build. 231 (2021). https://doi.org/10.1016/j.enbuild.2020.110555.
- [304] M. Abdul Mujeebu, F. Bano, Energy-saving potential and cost-effectiveness of active energy-efficiency measures for residential building in Warm-humid climate, Energy for Sustainable Development. 67 (2022) 163–176. https://doi.org/10.1016/j.esd.2022.01.011.
- [305] S. Barbosa, J. Carlo, K. Ip, Energy performance of PV integrated office buildings with fan-assisted double skin façades under tropical climates, Int J Green Energy. 16 (2019) 1061–1072. https://doi.org/10.1080/15435075.2019.1653879.
- [306] P. Prajongsan, S. Sharples, Enhancing natural ventilation, thermal comfort and energy savings in high-rise residential buildings in Bangkok through the use of ventilation shafts, Build Environ. 50 (2012) 104–113. https://doi.org/10.1016/j.buildenv.2011.10.020.
- [307] B. Raji, M.J. Tenpierik, R. Bokel, A. van den Dobbelsteen, Natural summer ventilation strategies for energy-saving in high-rise buildings: a case study in the Netherlands, International Journal of Ventilation. 19 (2020) 25–48. https://doi.org/10.1080/14733315.2018.1524210.
- [308] G. Elshafei, A. Negm, M. Bady, M. Suzuki, M.G. Ibrahim, Numerical and experimental investigations of the impacts of window parameters on indoor natural ventilation in a residential building, Energy Build. 141 (2017) 321–332. https://doi.org/10.1016/j.enbuild.2017.02.055.
- [309] M. Aram, O. Abessi, Optimal design of green buildings using computational fluid dynamics and climate simulation tools, International Journal of Environmental Science and Technology. 17 (2020) 917–932. https://doi.org/10.1007/s13762-019-02403-6.
- [310] E. Bay, A. Martinez-Molina, W.A. Dupont, Assessment of natural ventilation strategies in historical buildings in a hot and humid climate using energy and CFD simulations, Journal of Building Engineering. 51 (2022). https://doi.org/10.1016/j.jobe.2022.104287.
- [311] G.M. Abbas, I. Gursel Dino, The impact of natural ventilation on airborne biocontaminants: a study on COVID-19 dispersion in an open office, Engineering, Construction and Architectural Management. 29 (2022) 1609–1641. https://doi.org/10.1108/ECAM-12-2020-1047.
- [312] T. Liu, X. Wang, W.L. Lee, Evaluating the effectiveness of transom window in reducing cooling energy use in high-rise residential buildings in Hong Kong, Journal of Building Engineering. 35 (2021). https://doi.org/10.1016/j.jobe.2020.102007.
- [313] D.P. Albuquerque, N. Mateus, M. Avantaggiato, G. Carrilho da Graça, Full-scale measurement and validated simulation of cooling load reduction due to nighttime natural ventilation of a large atrium, Energy Build. 224 (2020). https://doi.org/10.1016/j.enbuild.2020.110233.
- [314] G. Carrilho Da Graça, N.R. Martins, C.S. Horta, Thermal and airflow simulation of a naturally ventilated shopping mall, Energy Build. 50 (2012) 177–188. https://doi.org/10.1016/j.enbuild.2012.03.037.
- [315] F.M. Baba, H. Ge, R. Zmeureanu, L.L. Wang, Optimizing overheating, lighting, and heating energy performances in Canadian school for climate change adaptation: Sensitivity analysis and multi-objective optimization methodology, Build Environ. 237 (2023). https://doi.org/10.1016/j.buildenv.2023.110336.
- [316] S.K. Gupta, P.R. Chanda, A. Biswas, A 2E, energy and environment performance of an optimized vernacular house for passive cooling Case of North-East India, Build Environ. 229 (2023). https://doi.org/10.1016/j.buildenv.2022.109909.
- [317] K. Grygierek, J. Ferdyn-Grygierek, Multi-objective optimization of the envelope of building with natural ventilation, Energies (Basel). 11 (2018). https://doi.org/10.3390/en11061383.

- [318] F. Harkouss, F. Fardoun, P.H. Biwole, Passive design optimization of low energy buildings in different climates, Energy. 165 (2018) 591–613. https://doi.org/10.1016/j.energy.2018.09.019.
- [319] J. Song, W. Wang, P. Ni, H. Zheng, Y. Zhou, Y. Zhang, Study on optimization method of summer nature ventilation for residential buildings in typical thermal zone of Xinjiang, China, Energy Reports. 8 (2022) 181–197. https://doi.org/10.1016/j.egyr.2022.05.086.
- [320] V. Pérez-Andreu, C. Aparicio-Fernández, A. Martínez-Ibernón, J.L. Vivancos, Impact of climate change on heating and cooling energy demand in a residential building in a Mediterranean climate, Energy. 165 (2018) 63–74. https://doi.org/10.1016/j.energy.2018.09.015.
- [321] D. Bienvenido-Huertas, D. Sánchez-García, C. Rubio-Bellido, J. Solís-Guzmán, Using adaptive strategies of natural ventilation with tolerances applied to the upper limit to improve social dwellings' thermal comfort in current and future scenarios, Sci Technol Built Environ. 28 (2022) 527–546. https://doi.org/10.1080/23744731.2022.2040884.
- [322] F. Fiorito, G. Vurro, F. Carlucci, L.M. Campagna, M. De Fino, S. Carlucci, F. Fatiguso, Adaptation of Users to Future Climate Conditions in Naturally Ventilated Historic Buildings: Effects on Indoor Comfort, Energies (Basel). 15 (2022). https://doi.org/10.3390/en15144984.
- [323] S. Thapa, Risk of overheating in low-rise naturally ventilated residential buildings of northeast India–an effect of climate change, Archit Sci Rev. 65 (2022) 14–41. https://doi.org/10.1080/00038628.2021.1941748.
- [324] L. Pajek, M. Košir, Strategy for achieving long-term energy efficiency of European single-family buildings through passive climate adaptation, Appl Energy. 297 (2021). https://doi.org/10.1016/j.apenergy.2021.117116.
- [325] S. Gilani, W. O'Brien, Natural ventilation usability under climate change in Canada and the United States, Building Research and Information. 49 (2021) 367–386. https://doi.org/10.1080/09613218.2020.1760775.
- [326] L. Wang, X. Liu, H. Brown, Prediction of the impacts of climate change on energy consumption for a medium-size office building with two climate models, Energy Build. 157 (2017) 218–226. https://doi.org/10.1016/j.enbuild.2017.01.007.
- [327] S. Flores-Larsen, C. Filippín, G. Barea, Impact of climate change on energy use and bioclimatic design of residential buildings in the 21st century in Argentina, Energy Build. 184 (2019) 216–229. https://doi.org/10.1016/j.enbuild.2018.12.015.